

Federal Reserve Bank of Chicago

Eat or Be Eaten: A Theory of Mergers and Firm Size

Gary Gorton, Matthias Kahl, and Richard Rosen

WP 2006-14

Eat or Be Eaten: A Theory of Mergers and Firm Size*

Gary Gorton The Wharton School, University of Pennsylvania and NBER

Matthias Kahl Kenan-Flagler Business School, University of North Carolina at Chapel Hill

> Richard Rosen Federal Reserve Bank of Chicago

This version: June 30, 2006

Keywords: Mergers, managerial motivations, defensive mergers. JEL classification: G34

Abstract

We propose a theory of mergers that combines managerial merger motives and a regime shift that may lead to some value-increasing merger opportunities. Anticipation of the regime shift can lead to mergers, either for defensive or positioning reasons. Defensive mergers occur when managers acquire other firms to avoid being acquired themselves. Mergers may also allow a firm to position itself as a more attractive takeover target and earn a takeover premium. The identity of acquirers and targets and the profitability of acquisitions depend, among other factors, on the distribution of firm sizes within an industry.

^{*} The views in this paper are those of the authors and may not represent the views of the Federal Reserve Bank of Chicago or the Federal Reserve System. We thank Amit Goyal, Cabray Haines, Shah Hussain, Feifei Li, and Yihui Wang for excellent research assistance. We are grateful for helpful comments and suggestions by Andres Almazan, Antonio Bernardo, Sanjai Bhagat, Bhagwan Chowdhry, Michael Fishman, Gunter Strobl, S. Viswanathan, Ivo Welch, seminar participants at the University of Houston, Purdue University, and the University of Wisconsin-Madison, and conference participants at the AFA 2000 meetings and the Texas Finance Festival 2000. This is a substantially revised version of an early draft that was presented at these conferences. Of course, all remaining errors and shortcomings are solely our responsibility.

Introduction

The 1990s produced the greatest wave of mergers in U.S. history. Between 1995 and 2000, U.S. merger volume set a new record every year, expanding from \$800 billion in 1995 to \$1.8 trillion in 2000.¹ Due to the growth and importance of mergers, a substantial academic literature has developed to examine them. However, existing merger theories remain unable to reconcile certain key facts about merger activity.

Two of the most important stylized facts about mergers are the following: First, the stock price of the acquirer in a merger decreases on average when the merger is announced.² Recent work shows that this result is driven by negative announcement returns for very large acquirers, while small acquirers tend to gain in acquisitions (Moeller et al. (2004) and Kahl and Rosen (2002)). Second, mergers concentrate in industries that have experienced regime shifts in technology or regulation. Mergers may provide an efficient strategy for managers coping with such a shift and seeking to maximize the value of their firms (see, for example, Mitchell and Mulherin (1996), Andrade, Mitchell, and Stafford (2001), and Andrade and Stafford (2004)).

The view that mergers are an efficient response to regime shifts by value-maximizing managers – the so-called neoclassical merger theory (see, for example, Mitchell and Mulherin (1996), Weston, Chung, and Siu (1998), and Jovanovic and Rousseau (2002)) - can explain the second stylized fact. However, it has difficulties explaining negative abnormal returns to acquirers. Theories based on managerial self-interest or a desire for larger firm size and diversification (for example, Morck, Shleifer, and Vishny (1990)) can explain negative acquirer returns.³ However, they cannot explain why mergers are concentrated in industries undergoing a regime shift.

This paper provides a theory of mergers that combines elements from both of these schools

¹ See "The Year of the Mega Merger...", Fortune Magazine, January 11, 1999, "Tales of the Tape: '99 M&A Vol Hits Record...", Dow Jones News Service, December 29, 1999, and "Year-End Review of Markets & Finance 2000..", The Wall Street Journal, January 2, 2001.

² Studies that find negative average returns to bidders include Asquith, Bruner, and Mullins (1987), Banerjee and Owers (1992), Bradley, Desai, and Kim (1988), Byrd and Hickman (1992), Jennings and Mazzeo (1991), Servaes (1991), Varaiya and Ferris (1987), and You, Caves, Smith, and Henry (1986). See Table 8-6 in Gilson and Black (1995). See also the recent survey by Andrade, Mitchell, and Stafford (2001). Bradley and Sundaram (2004) find, using a much larger sample of mergers in the 1990s, that most acquirers experience positive announcement returns. The negative announcement returns are concentrated among stock financed acquisitions of public targets that are large relative to the acquirer.

³ Other papers in which managerial motivations for mergers are prominent include Amihud and Lev (1981), Shleifer and Vishny (1989), and May (1995). Theoretical papers representing the neoclassical tradition include Gort (1969) and Rubin (1973). Negative acquirer announcement returns can be explained without appealing to agency conflicts between managers and owners if the takeover announcement reveals negative information about the acquirer's profitability relative to expectations (see McCardle and Viswanathan (1994) and Jovanovic and Braguinsky (2004)).

of thought. The notion of a regime shift that *may* make mergers an efficient choice for managers remains a key part of our analysis. However, one particular managerial motive – the desire not to be acquired – is also important. Our theory of mergers is able to reconcile the above two stylized facts and also explain a third important characteristic of mergers – the fact that they often come in waves. Firm size plays a key role in our theory, which can help explain the differing findings on acquirer returns for small and large firms.

The basic elements of our theory are the following: First, we assume that managers derive a private benefit from operating a firm in addition to the value of any ownership share of the firm they have. This means that self-interested managers may have a preference for keeping their firms independent since managers of acquired firms may lose private benefits because they are likely to either play subordinated roles in the new firms or lose their jobs (Morck, Shleifer, and Vishny (1988)).

Second, we assume that there is the possibility of a regime change that creates economies of scale (synergies). After the regime shift, some mergers create value, and larger targets are more attractive merger partners due to economies of scale. As a consequence, firm size is an important determinant of which firms are takeover targets. We assume that the regime shift is uncertain. *Ex ante*, the probability of the regime shift is small enough that mergers do not create expected synergies.

Third, we assume that a firm can only acquire a smaller firm. While this is an assumption in our model, there are reasons why we rarely see firms acquire larger rivals. For one, a larger acquisition is more difficult to finance. Typically, it will be more difficult to raise funds by issuing debt for a larger acquisition. Adding a lot of debt can also substantially increase the chance of financial distress, and managers of financially distressed firms are more likely to lose their jobs (see Gilson (1989)). Alternatively, acquiring a larger company with stock would dilute the acquirer's ownership of the combined company and perhaps lead to a loss of control for incumbent management. These difficulties in acquiring larger companies may explain why in most mergers the acquirer is considerably larger than the target and the probability of being a target is decreasing in firm size (e.g., see Hasbrouck (1985) and Palepu (1986)).

In our model, the anticipation of potential mergers after the regime shift creates incentives to engage in additional mergers. We show that a race to increase firm size through mergers can ensue for either defensive or positioning reasons. Defensive mergers occur because when managers care sufficiently about staying in control, they may want to acquire other firms to avoid being acquired themselves. By growing larger through acquisition, a firm is less likely to be acquired itself as it becomes bigger than some rivals. This defensive merger motive is self-reinforcing: One firm's defensive acquisition makes other firms more vulnerable as takeover targets, which induces them to make defensive acquisitions themselves. This leads to an "eat-or-be-eaten" scenario,⁴ in which unprofitable defensive acquisitions preempt some or all profitable acquisitions. We show that in industries in which many firms are of similar size to the largest firm, defensive mergers are likely to occur.

Besides defensive motives, there is another reason why the anticipation of efficient mergers can lead to a race for firm size. Since following the regime shift, merger synergies are increasing in size (due to economies of scale), by becoming larger, a firm can become a more attractive takeover target. Being acquired generates a takeover premium for the target and for its manager, who owns a share of the firm. We show that in industries in which there is a dominant firm, such "positioning" mergers are likely. In these industries, no merger ensures that a firm becomes large enough so that it cannot be acquired by the largest firm. Indeed, acquiring another firm has the opposite effect of making the firm a more attractive takeover target. If managers care enough about preserving the independence of their firms, they *avoid* acquisitions. But, if managers care a lot about firm value, that is, private benefits are low, they may engage in acquisitions in order to position themselves as a *more* attractive target. All acquisitions are profitable, because the early acquisitions are undertaken to increase the likelihood of being the target in a wealth-creating merger later. Here, merger waves occur only if managers care sufficiently about maximizing firm value – in contrast to the case of industries in which all firms are of similar size, where waves occur when managers care more about private benefits.

We also consider an industry structure where both defensive and positioning mergers are possible. In an industry in which some but not all firms are of a similar size, medium-size firms have both the opportunity to make defensive acquisitions (that make them large enough to be protected from takeovers) as well as positioning acquisitions (that make them more attractive takeover targets). In these industries, the pattern of mergers depends crucially on firm size and the level of managerial private benefits. We show that the profitability of acquisitions is generally decreasing in the acquirer's size. Large firms engage only in defensive, unprofitable acquisitions,

⁴ Many articles in the press or trade journals mention the idea that if firms do not make acquisitions, they may become targets themselves. Often, they refer to the ensuing merger dynamics as "eat or be eaten" (see, for example, "For Military Contractors, It's Buy or Be Bought", New York Times, March 12, 1994, "Phone Giants Reportedly Mulling Merger", Los Angeles Times, May 11, 1998, and "Middle-market M&A players: Where are they now?", Bank Mergers & Acquisitions, August 1, 2002). Sometimes these articles explicitly discuss the role that firm size can play in who can acquire whom (see, for example, "New England Utility Mergers Raise Question over Boston Edison's Future", KRTBN Knight-Ridder Tribune Business News: The Boston Globe – Massachusetts, June 16, 1999, and "Takeover Rumors Fuel Another Rise in Amgen Stock…", The Los Angeles Times, August 30, 1994).

and these only occur when private benefits are high.⁵ Medium-size firms engage in unprofitable defensive acquisitions when private benefits are high, but when private benefits are low, they engage in profitable positioning acquisitions. The profitability of their acquisitions decreases in the size of the target relative to that of the acquirer. Finally, small firms typically engage in profitable acquisitions.⁶ These mixed firm size industries are most likely to exhibit merger waves, because some firms have defensive as well as positioning merger motives. Which motive matters depends on their managers' interest in maximizing firm value. While the additional acquisitions may not be undertaken by medium-size firms, those firms tend to make the most acquisitions.

Overall, our models show that in industries with economies of scale, firm size becomes the driving force for merger dynamics. Often, this leads to profitable acquisitions. However, if a firm becomes very large and its manager's private benefits are high, it may engage in an unprofitable defensive acquisition.

Our theory can explain the three stylized facts about mergers mentioned above. It also generates a number of other empirical predictions. For example, we predict that (1) acquirer returns are negatively correlated with acquirer size as well as target size (consistent with the results in Kahl and Rosen (2002), Moeller, Schlingemann, and Stulz (2004), and Kahl and Valkanov (2004)); (2) that acquirer announcement returns for medium-size firms decrease in the ratio of target to acquirer size; and that (3) medium-size firms are most likely to acquire other firms. However, we caution that the notion of size in our model – size relative to the largest firm in the industry with a profitable acquisition opportunity - is slightly different than absolute size. These and other empirical implications are discussed in more detail in section V.

Another implication of our model that is central to our theory is that the firm size distribution in an industry matters for merger dynamics. To our knowledge, other merger theories do not have strong predictions about how acquisitions are related to industry structure. In particular, our model predicts that firms in industries with more medium-size firms have a higher probability of making acquisitions. We use data on U.S. mergers from 1983-2003 to test this hypothesis, and we find support for it.

Our paper is related to several other papers. Harris (1994) presents a model that determines which firm is the acquirer and which firm the target in a merger that is always value-increasing. She also assumes that managers have a preference for being the acquirer rather than the target in an acquisition. Her model is static and has only two firms. In contrast, we analyze merger dynam-

⁵ In our model, the largest firm makes only profitable acquisitions, because it has no defensive motives. However, we view this firm as a modeling device.

⁶ However, there is one unprofitable acquisition by a small firm for very high private benefits in our model.

ics involving several firms. This allows us to generate results on the timing of mergers, on merger waves, and on the effect of the distribution of firm sizes in an industry on the merger dynamics, the identity of acquirers and targets, and the profitability of acquisitions. Toxvaerd (2004) analyzes strategic merger waves. Acquirers compete over time for a scarce set of targets. The trade-off between the option value of waiting for better market conditions and the fear of being pre-empted by other acquirers and hence being left without a merger partner determines the merger timing. This can give rise to strategic merger waves. Managerial acquisition incentives or the importance of size as a takeover deterrent do not play a role in the analysis.

There are a number of other important papers on mergers. Shleifer and Vishny (2003) argue that merger waves are driven by misvaluations in the stock market.⁷ Rhodes-Kropf and Viswanathan (2004) show that even in a rational model with efficient markets stock market booms can lead to merger waves. Roll (1986) argues that acquirers may overpay because managers may fall prey to hubris.⁸ Lambrecht (2004) and Morellec and Zhdanov (2005) emphasize the real options aspects of merger decisions. While we abstract from all these issues, we do not mean to suggest that they do not have an important impact on mergers.

Our paper is also related to the Industrial Organization literature on mergers (e.g., Berry and Pakes (1993), Baker (1997), Gowrisankaran (1999), Epstein and Rubinfeld (2001), and Peters (2003)).⁹ This literature illustrates many of the modeling issues that must be confronted in analyzing mergers. In order to make the models tractable and reduce the potentially large multiplicity of equilibria, one has to make strong assumptions concerning the assumed choice of merger decision protocol (e.g., the order in which firms decide on whether to merge), initial industry size distribution, division of any surplus between merging entities, and so on. We also have to make strong assumptions on the merger protocol to keep our models tractable.

The remainder of the paper is structured as follows: Section I presents the basic model and an extension to merger waves in an industry with firms of similar size. Section II offers a model with a dominant firm. Section III presents a model in which some but not all firms are of similar size. Section IV discusses some of the assumptions of our models. Section V presents some addi-

This acquisition secures the independence of the small firm for one, but not two, periods.

⁷ For empirical evidence supporting this theory see Ang and Cheng (2003), Dong, Hirshleifer, Richardson, and Teoh (2003), and Rhodes-Kropf, Robinson, and Viswanathan (2005). However, Harford (2005) finds that industry shocks and sufficient liquidity in the capital markets are more important in causing merger waves than market timing attempts by managers.

⁸ For empirical evidence supporting the hubris hypothesis, see, for example, Rau and Vermaelen (1998) and Malmendier and Tate (2003).

⁹ These models typically cannot be solved analytically. However, Gowrisankaran and Holmes (2004) analytically analyze the effect of mergers on industry concentration.

tional empirical implications of our theory. Section VI offers a short empirical part. Section VII concludes.

I. The Basic Model

This section presents the basic model. It has three firms, the minimum number needed to generate interesting results. The other models below are variations of this model and have the same basic structure but are more complicated and involve more firms. The basic model is rich enough to allow mergers where managers have a better chance of retaining private benefits (in our model, equivalent to keeping their jobs) at the expense of shareholder value. We call these defensive mergers. This model does not have a role for value-increasing positioning mergers, which are undertaken to increase the chance of the firm being taken over and hence earning a premium. A model of positioning mergers needs at least four firms and will be presented in the next section.

A. Model Assumptions

The simplest way to model managerial motivations for mergers is using two dates, 0 and 1, and three firms. The timing of the model is shown in Figure 1.

The three firms are ordered by their size (stand-alone value) C_i , with $C_1 > C_2 > C_3$, so Firm 1 is the largest and Firm 3 the smallest (we always assume that firms are ordered by size, with Firm *i* larger than Firm *j* if i < j).¹⁰ We assume that firms cannot acquire firms that are larger than they are, implying that every firm is immune against acquisition attempts by firms that are smaller than it is. Hence, the larger a firm is, the fewer potential acquirers it has. As we discussed in the Introduction, there are probably fewer buyers for larger firms since acquiring a larger firm requires more resources and is more likely to lead to a loss of control.¹¹ As a consequence of the assumption that a firm cannot acquire a larger firm, Firm 3 cannot acquire another firm, because it is the smallest firm. To make the model interesting, we assume that $C_2 + C_3 > C_1$, i.e., that after Firm 2 acquires Firm 3, it is larger than Firm 1 and hence cannot be acquired. This gives Firm 2 a potential defensive incentive to acquire Firm 3.

¹⁰ In interpreting our models, it should be recognized that antitrust laws put some restrictions on intraindustry or horizontal mergers. This is one reason why the largest firm with a profitable acquisition opportunity (Firm 1) may not necessarily be the largest firm in the industry. Our model considers mergers within an industry that are allowed by antitrust law.

¹¹ As mentioned in the Introduction, it has been found that the probability of being a target in an acquisition is decreasing in a firm's size (see, e.g., Hasbrouck (1985) and Palepu (1986)).

At each date, a manager receives private benefits of w if his firm is not acquired and zero if it is acquired.¹² The manager of each firm also owns a share α of his firm, which, for simplicity, is exogenous. All firms resolve indifference between acquiring and not acquiring in favor of not acquiring, perhaps due to unmodeled transactions costs. We assume that contracts cannot fully overcome managerial preference for their firms to remain independent.¹³

We assume that at each date, a firm can make at most one acquisition offer to another firm. Within each period, Firm 1 moves first, Firm 2 moves second, and Firm 3 moves last. The profitability of a merger depends on the identities of the merger partners and on the state of nature, realized at date 1. Each firm can make at most one acquisition over the two dates: If it has made an acquisition at date 0, it cannot make another one at date 1. This assumption simplifies the analysis and is discussed in more detail in section IV below.

At the start of date 0 firms learn that a regime shift may occur, corresponding to a change in their environment. For example, a fundamental change in technology or government regulation may at some point in the (near) future make acquisitions profitable. The change may generate economies of scale or scope, or it may lead to overcapacity. The key is that the regime shift changes conditions in the industry enough to make some acquisitions possibly profitable, but the probability of this regime shift is low enough such that current acquisitions are not yet profitable for the acquirer.

More concretely, we assume that mergers at date 1 are *possibly* efficient. Specifically, firms learn at date 0 that the state of nature at date 1 will be good with probability ρ and bad with probability $1 - \rho$. In the good state of the world, only Firm 1 has efficient merger opportunities in the sense that if Firm 2 or Firm 3 is combined with Firm 1, it is worth more than its stand-alone value of C₂ or C₃, respectively, perhaps due to synergies (economies of scale and scope).

Without loss of generality, we assume that if the good state occurs at date 1, Firm 2 and Firm 3 are worth twice as much when combined with Firm 1 than as stand-alones: Firm 2 is worth C_2

 $^{^{12}}$ In reality, private benefits differ across firms and managers and arguably increase in firm size. In a model that captured this cross-sectional variation in private benefits, firms with larger private benefits would be more inclined to engage in defensive acquisitions. Moreover, we conjecture that unprofitable acquisitions would be even more attractive to managers since the higher private benefits that can be obtained in larger firms would give them another reason to make acquisitions in their own interests at the expense of their shareholders. Our assumption of homogenous private benefits allows us to spell out the consequences of one particular managerial motive – survival of the firm as an independent entity – while abstracting from other motives such as general empire-building tendencies that are already well understood in the literature (Baumol (1959), Marris (1964), Jensen (1986), and Jensen (1993)).

¹³ An assumption that the agency conflict cannot be fully eliminated through compensation contracts is typically made in the literature (see, e.g., Hart and Moore (1995)).

as a stand-alone, but is worth $2C_2$ if combined with Firm 1. Similarly, Firm 3 is worth C_3 as a stand-alone, but is worth $2C_3$ if combined with Firm 1. However, if Firm 2 combines with Firm 3, there are no synergies: their combined value is equal to the sum of their stand-alone values $C_2 + C_3$. Assuming that a combination of Firms 2 and 3 would create value would not change the main results.¹⁴

In the bad state at date 1, all mergers destroy value. For simplicity, we assume that any target acquired by another firm is worth zero after the merger. Again, this is just a normalization, and it could easily be adjusted without affecting the qualitative results of our analysis.

To make the merger decision interesting, we want mergers to be unprofitable (valuereducing), even for Firm 1 at date 0. This will be the case if the bad state is more likely than the good state:

(1) $\rho < 0.5$.

Firms might make unprofitable acquisitions in our model because we assume that the acquiring firm's *managers* decide whether or not to make an offer. We concentrate on the motivations to make acquisition offers, but we also have to think about when a target firm will accept an offer. Assume that *shareholders* of the target firm make the decision about whether to accept an offer.¹⁵ We assume that target shareholders accept a takeover offer if and only if it involves at least a zero premium over its stand-alone value.¹⁶

¹⁴ We have verified that the equilibrium merger dynamics would be very similar if the value of Firm 3 doubled when combined with Firm 2 in the good state at date 1. The only difference would be that the cut-off level of private benefits above which Firm 2 acquires Firm 3 at date 0 would be lower, because the date 0 acquisition is now less unprofitable.

¹⁵ A central idea behind our model is that takeovers can serve a defensive purpose in that an acquisition reduces the probability of being acquired. Hence, defensive acquisitions can be viewed as takeover defenses. Our model ignores other takeover defenses such as poison pills and staggered boards. To the extent that these other takeover defenses can be employed, they may substitute for defensive acquisitions and hence make defensive acquisitions less relevant. We agree that other takeover defenses can be important and effective (see, for example, Bebchuk, Coates, and Subramanian (2002)). In the literature, the effectiveness of takeover defenses is still being debated. For example, Ambrose and Megginson (1992) find that the only common takeover defense that is significantly negatively correlated with acquisition likelihood is blank-check preferred stock authorizations. Comment and Schwert (1995) conclude that "poison pill rights issues, control share laws, and business combination laws have not systematically deterred takeover measures increase the bargaining position of target firms, but they do not prevent many transactions." (p.3). See also Jensen and Ruback (1983), Jarrell, Brickley and Netter (1988), and Ruback (1988) for earlier surveys of the evidence on takeover defenses. Also see Schwert (2000).

¹⁶ This assumption simplifies the analysis considerably, because we do not have to analyze whether the target firm is interested in rejecting an offer in the hope of a better offer by the same firm or another firm later on. We conjecture that our main results are robust to relaxing this assumption.

Finally, we determine the price at which a firm can acquire another firm by applying the Nash bargaining solution to a bargaining game between the acquiring firm's manager and the target firm's shareholders. This implies that they equally share the surplus from the merger in terms of an increase in utility for the acquiring firm's manager and the premium to the target firm's shareholders. We ignore the target firm's manager in the bargaining since we assume that target shareholders, not the manager, accept or reject a takeover offer. Of course, as discussed before, if the target firm's manager could prevent a takeover, there would be no incentives to make defensive acquisitions. We ignore the acquiring firm's shareholders in the bargaining game, since we assume that it is the acquiring firm's manager, not the shareholders, who make takeover decisions. If the acquiring firm's shareholders could prevent a merger if it is not in their interests, there would be no unprofitable acquisitions.

The Nash bargaining solution implies the following characterization of the premium the acquirer pays over the stand-alone value of the target: If Firm A acquires Firm B, it pays a premium above the stand-alone value of Firm B such that Firm A's manager retains one half of his utility gain that arises from the acquisition if the premium were zero. Put differently, the premium is determined such that one half of the difference between the utility of Firm A's manager if he acquires Firm B at a zero premium and his utility if he does not acquire Firm B but subsequently behaves optimally (perhaps acquiring Firm B at a later point of time) accrues to Firm A's manager. For example, suppose Firm 1 acquires Firm 2 in the good state at date 1. Because Firm 2 is worth $2C_2$ if combined with Firm 1 but Firm 1 pays only C_2 for it if there is no premium, Firm

1 gains C₂ and its manager gains α C₂. If Firm 1 pays C₂ + 0.5 $\frac{1}{\alpha}\alpha$ C₂ = 1.5C₂ for Firm 2, Firm

1's manager gains $0.5\alpha C_2$ from the acquisition, which is half of what he would gain without paying a premium. We use this price below. Of course, the premium above stand-alone value has to be nonnegative. Otherwise, Firm B's shareholders will not agree to the acquisition. Given these assumptions, an acquisition occurs whenever it is in the interest of the acquirer's manager if he offers at least a zero premium over the target's stand-alone value. Note that the outside opportunity for the acquirer as well as the target depends on the future (optimal) merger activity. For example, if Firm 2 acquires Firm 3 at date 0, the premium it pays over Firm 3's stand-alone value, C_3 , reflects that in the absence of this acquisition, Firm 2 is acquired by Firm 1 but Firm 3 remains independent. Hence, the premium takes into account expected merger activity.

Note also that this bargaining model implies that acquirers may pay a positive premium even though there are no (or even negative) synergies between acquirer and target. Indeed, such mergers could not occur if the premium did not reflect to some extent the private benefits gained by the acquiring firm's manager. Otherwise, the premium would have to be negative, and that would not be accepted by the target's shareholders. Hence, the target extracts to some extent the acquiring firm's private benefits from the acquisition, via the premium. Otherwise, negative synergies mergers could not occur. More generally, unprofitable acquisitions (even if there are positive synergies) can occur only if the acquiring firm's manager pays a premium higher than the synergies and hence shares some of the private benefits he gains with the target firm's shareholders.

B. Equilibrium Merger Activity

To find the equilibrium pattern of mergers, we solve the model by backwards induction, starting at date 1.¹⁷ Firm 2 moves last (since Firm 3 has no strategic decision to make). If the realized state of nature at date 1 is bad, all acquisitions are unprofitable. Firm 2 will not acquire Firm 3 since the acquisition destroys value but Firm 2 has to pay at least the stand-alone value of the target firm. Moreover, Firm 2 does not need to defend itself against a potential acquisition, because Firm 1 does not have an opportunity to acquire it anymore. Firm 1 will not acquire Firm 2 or Firm 3 since the acquisition is unprofitable and Firm 1, being the largest firm, does not need to defend itself against a potential acquisition.

If the state at date 1 is good, Firm 1 will acquire the largest remaining firm, Firm 2, because its gains from the merger are increasing in the size of the target. If Firm 2 has not acquired Firm 3 before, Firm 1 acquires it and its manager gains up to αC_2 , his share of the synergies from the merger. Hence, he pays $1.5C_2$ for Firm 2, as noted earlier. Firm 2, because it is acquired by Firm 1, does not get to make an acquisition offer.

Now let us turn to date 0. Firm 2 can acquire Firm 3. While this merger reduces shareholder value, Firm 2's manager may want to acquire Firm 3 since this ensures that he keeps his private benefits at date 1. If Firm 2 acquires Firm 3 at date 0 and pays a zero premium, the expected utility of Firm 2's manager is $2w - \alpha(1-\rho)C_3$. The manager is employed in both dates, so he gets total private benefits of 2w. The merger with Firm 3 generates no synergies in the good state at date 1, but destroys C_3 in value in the bad state. Since the manager owns a share α of Firm 2 and the bad state occurs with probability $1-\rho$, the manager has an expected utility loss of $\alpha(1-\rho)C_3$. On the other hand, if Firm 2 does not acquire Firm 3 at date 0, it will be acquired by Firm 1 in the good state at date 1, as seen above. Hence, the expected payoff of Firm 2's manager if he does not acquire Firm 3 at date 0 is $w + (1-\rho)w + \rho 0.5\alpha C_2$, equal to the private benefits at

¹⁷ The unique subgame perfect equilibrium outcome can be found by backwards induction, because our model is a game of complete and perfect information.

date 0 and in the bad state at date 1 plus the premium paid to the manager as a part owner of the firm when it is acquired by Firm 1 in the good state at date 1. Note that this calculation takes into account that the outside opportunity for Firm 2's manager involves being acquired by Firm 1 in the good state at date 1 at a premium. It can be calculated that Firm 2 chooses to acquire Firm 3 at date 0 if and only if:

(2) w > 0.5
$$\alpha$$
C₂ + $\alpha \frac{1-\rho}{\rho}$ C₃.

Firm 1 never acquires another firm at date 0. Such an acquisition will be unprofitable.¹⁸ Proposition 1 summarizes the above discussion.

Proposition 1: If private benefits are low enough so that (2) does not hold, Firm 1 acquires Firm 2 in the good state at date 1 in a profitable acquisition. If private benefits are high enough so that (2) holds, Firm 2 acquires Firm 3 at date 0 in an unprofitable acquisition.

Hence, for low private benefits, there is only the profitable (positive NPV) acquisition of Firm 2 by Firm 1. It is profitable because there are positive synergies (Firm 2 is worth more if combined with Firm 1) and Firm 1 appropriates some of this gain. For high private benefits, Firm 2 acquires Firm 3 in an unprofitable (negative NPV) acquisition at date 0.

As can be seen from condition (2), for a given level of private benefits, Firm 2 is less likely to acquire Firm 3 if α is larger, because then Firm 2's manager cares more about firm value.¹⁹ Moreover, if the probability of the good state at date 1 is higher, the minimal private benefits that induce the manager to engage in the date 0 acquisition are smaller. This implies that defensive mergers are more likely after a regime shift that increases the likelihood that future mergers are efficient. There are two reasons for this result. First, a date 0 acquisition becomes more attractive if the probability that such an acquisition turns out to not destroy value (the probability of the good state) increases. Second, if the probability of the good state is higher, there is a higher

¹⁸ Note also that there is no incentive for Firm 1 to acquire any other firm in order to improve its acquisition opportunities at date 1 since we have assumed that each firm can make only one acquisition. For example, Firm 1 may want to acquire Firm 3 at date 0 although this is unprofitable because it ensures that Firm 2 will be available as a target at date 1. However, because Firm 1 can only acquire one firm, it is never interested in acquiring Firm 3 at date 0 in order to make sure that it can acquire Firm 2 at date 1. The simplifying assumption that each firm can make only one acquisition is discussed in more detail in section IV, where we conjecture what would happen if we relaxed this assumption.

¹⁹ Consistent with our prediction, Lewellen, Loderer, and Rosenfeld (1985) find that managerial stock ownership has a positive impact on the quality of acquisitions managers make. Agrawal and Mandelker (1987) find that the smaller is the ownership stake of top managers (in the form of shares and options), the more likely are any mergers or sell-offs undertaken to reduce the variance of returns on the firm's assets or to

chance that Firm 2 will be acquired by Firm 1 if Firm 2 does not acquire Firm 3 at date 0. But since Firm 2's manager wants to avoid being acquired, he is more likely to be interested in acquiring Firm 3 at date 0 if his firm is more likely to be acquired otherwise.

Proposition 1 shows how managerial self-interest leads to unprofitable acquisitions. Moreover, when managers care about private benefits, there are more mergers than occur in the firstbest world (with no managerial merger motives). This is because managers do not just acquire when they are otherwise certain to be acquired, but they also acquire even if there is a significant chance of being acquired in the future. In our model, the expected number of efficient mergers is ρ , while the actual number for sufficiently high private benefits is one.

Note that the premium over stand-alone value that is paid by Firm 2 in the date 0 acquisition of Firm 3 $(0.5\frac{1}{\alpha}(\rho_W - \alpha(1-\rho)C_3 - 0.5\rho\alpha C_2))$ can be higher than the premium over stand-alone value that Firm 1 pays for Firm 2 in the good state at date 1, even though the latter is an efficient, value-creating merger while the former is value-destroying. The reason is that when Firm 2 acquires Firm 3, Firm 3 extracts some of the private benefits that Firm 2's manager preserves through the acquisition. However, the premium that Firm 1 pays for Firm 2 does not involve any private benefits, since the merger is not motivated by the preservation of any private benefits. Hence, if private benefits are sufficiently high, the inefficient acquisition of Firm 3 by Firm 2 can be associated with a larger premium. This is despite the fact that the premium that Firm 2 pays for Firm 3 at date 0 takes into account that in the absence of this acquisition Firm 2 is acquired by Firm 3 in the good state at date 1 at a positive premium – reducing the premium Firm 2 pays for Firm 3.

C. Defensive Merger Waves

In this section, we show how defensive motivations can lead to merger waves, where we define a merger wave as an equilibrium with more than one merger. For this, we expand the basic model in the previous section to five firms. Merger synergies are similar to those in the three-firm model. In the good state at date 1, if Firm j is combined with Firm 1, its value is $2C_j$ while all other combinations lead to neither positive nor negative synergies. We assume that only one firm has profitable acquisition opportunities (arising from positive synergies) to show that even if this is the case, our model can generate more than one merger. We assume that Firm 1 is the firm with the profitable acquisition opportunity, because this generates defensive merger motives for all

reduce its leverage - goals that are in the managers', but not the shareholders' interests.

other firms. Since firms that are larger than the largest firm with a profitable acquisition opportunity never participate in any merger, we can ignore such firms. Hence, giving Firm 1 the profitable acquisition opportunity is without loss of generality. As before, we assume that any firm that is acquired is worth zero in the bad state at date 1 and that each firm can at each date make only one acquisition offer to another firm (and, again, make only one acquisition in total over both dates). So, our assumptions mean that the efficient number of mergers is one and the maximum number of mergers is two.

We assume that all firms are sufficiently close in size such that the acquisition of any other firm makes each firm larger than Firm 1: $C_4 + C_5 > C_1$. We refer to this industry as a homogenous firm size industry. Further, assume that if a firm is indifferent between acquiring two firms, it resolves this indifference in favor of acquiring the smaller of the two firms. Finally, we assume that within each period, Firm 4 moves first, Firm 3 second, Firm 2 third, and Firm 1 last (Firm 5 has no strategic decisions to make). We reverse the order of moves relative to the three-firm model in the previous section, because this makes the intuition behind merger waves richer. However, if we assumed the opposite (original) order of moves, we would get similar results – and in particular, we would also get defensive merger waves.

C.1. Analysis

Proposition 2 summarizes the merger activity for all parameter regions.²⁰ The following condition plays an important role:

(3) $w \le 0.5 \alpha C_2$

Proposition 2: If private benefits are low enough so that (3) holds there is only one acquisition: Firm 1 acquires Firm 2 in the good state at date 1. This acquisition is profitable. If private benefits are high enough so that (3) does not hold, there are two acquisitions and both of them are unprofitable: Firm 4 acquires Firm 5 and Firm 2 acquires Firm 3. The date of the mergers depends on private benefit levels. For intermediate values of private benefits, Firm 4 acquires Firm 5 at date 1 and Firm 2 acquires Firm 3 in the good state at date 1. For high values of private benefits, both of these acquisitions occur at date 0.

The Proposition shows that if private benefits are low, the only merger is the most efficient one between Firm 1 and Firm 2. No firm engages in a defensive acquisition because it is unprofit-

²⁰ All proofs, including for Proposition 2, are available from the authors upon request.

able and it also eliminates the chance of earning a takeover premium. However, if managers care enough about their private benefits so that (3) is not satisfied, the merger dynamics are drastically transformed from a world with only one profitable acquisition into a world with two defensive, unprofitable acquisitions, meaning that the efficient and profitable acquisition is preempted. When this happens, our model gives rise to a merger wave – two mergers either both at date 0, both at date 1, or one merger each at date 0 and date 1.

The intuition behind the merger wave is as follows. If no merger has occurred yet, Firm 1 acquires Firm 2 in the good state at date 1. Anticipating that, Firm 2 wants to acquire Firm 5 in the good state at date 1. This means that Firms 3 and 4 also want to acquire Firm 5 in the good state at date 1 to avoid being the largest remaining firm except Firm 1, and hence the most attractive target for Firm 1. But Firm 2 anticipates that and wants to preempt them by acquiring Firm 5 at date 0^{21} , leading Firm 3 and Firm 4 to have a similar incentive.²² Given the order of moves, Firm 4 acquires Firm 5. After that acquisition, Firm 2 can only secure its independence by acquiring Firm 3.

We also see from Proposition 2 that the mergers tend to occur earlier if private benefits are higher. In our model, date 0 mergers mean that the synergies are lower and hence total value destruction for the acquirer and the target combined is higher.²³ However, early acquisitions (at date 0) can be more profitable for the acquirer than late (date 1) acquisitions.²⁴ For example, if Firm 4 acquires Firm 5 at date 0 and then Firm 2 acquires Firm 3 at date 1, the first acquisition can be more profitable than the second. Although synergies are lower (because negative) in the first merger, Firm 2 may pay a higher premium in the date 1 acquisition than Firm 4 in the date 0 acquisition. The premium Firm 2 pays can be higher, because it partially transfers its manager's utility gain from staying in control for sure (since the acquisition ensures he will stay in control) to Firm 3. In contrast, the date 0 acquisition of Firm 5 by Firm 4 transfers partially the utility gain

 $^{^{21}}$ If it did not acquire Firm 5 and Firm 4 acquired Firm 5 in the good state at date 1, Firm 2 could still acquire Firm 3 in the good state at date 1, but that turns out to be less attractive than acquiring Firm 5 at date 0.

²² If Firm 3 did not acquire Firm 5, it could still acquire Firm 4 after Firm 2 acquired Firm 5, but this turns out to be less attractive.

 $^{^{23}}$ Even after Firm 4 has acquired Firm 5 at date 0, Firm 2 prefers acquiring Firm 3 at date 0 rather than at date 1 for high enough private benefits. The reason is that if Firm 2 acquires Firm 3 at date 1, this is its last chance to secure its independence. Firm 3 understands that and extracts part of the private benefits for itself in the form of a high takeover premium. If Firm 2 acquires Firm 3 at date 0, the only surplus that Firm 3 can extract from Firm 2 is the difference between its manager's utility if he acquires Firm 3 at date 0 as compared to date 1. Since both acquisitions secure Firm 2's independence, Firm 3 cannot extract the private benefits in a date 0 acquisition.

²⁴ The profitability of an acquisition is calculated as the change in the acquirer's payoffs arising from the acquisition divided by the acquirer's stand-alone value.

of Firm 4's manager from staying in control with only the probability that the good state arises (since in the bad state at date 1 he stays in control even without the acquisition). Hence, for high enough private benefits and low enough probability of the good state arising, the premium in the date 1 acquisition is sufficiently higher to outweigh the higher (not negative) synergies.

II. Positioning Mergers: A Model with a Dominant Firm

In the previous section we analyzed the merger dynamics in a situation in which all firms were of similar size so that even if the smallest two of the five firms merged, they became larger than the largest firm. We showed that in such an industry, defensive mergers are likely. In this section, we turn to a very different industry structure – one in which the largest firm is much larger than all the other firms. We show that in such an industry, the merger dynamics are very different. In particular, firms may undertake an acquisition to become more attractive targets for other firms. We call these positioning mergers. We present again the simplest model that generates the basic insights. For this purpose, a model with four firms suffices.

A. The Model

Assume there are four firms, with Firm 1 much larger than the other firms. In particular, let $C_1 > C_2 + C_3$. The other three firms are of a similar size so that Firm 3 and Firm 4 are, if combined, larger than Firm 2: $C_3 + C_4 > C_2$. We refer to this industry as one with heterogeneous firm size. The order of moves is as in the basic model with three firms. At each date, Firm 1 moves first, Firm 2 second, and Firm 3 last (since Firm 4 has no strategic decision to make). Merger synergies are as in the previous two models, with only acquisitions by Firm 1 in the good state of date 1 having positive synergies and all mergers in the bad state at date 1 having negative synergies.²⁵

B. Analysis

As in the previous section, we solve the model by backwards induction. In the bad state at date 1, there is no acquisition. Firm 1 does not acquire any other firm, and no other firm has an incentive to make an acquisition either. In the good state at date 1, Firm 1 acquires the largest

 $^{^{25}}$ We have verified that the equilibrium merger dynamics would be very similar if the target firm in an acquisition would be worth twice its stand-alone value in the good state at date 1 regardless of which firm is the acquirer (and hence also if the acquirer is Firm 2, 3, or 4). The only differences would be that the cutoff value for private benefits below which Firm 2 acquires Firm 4 at date 0 would be slightly different and that Firm 3 would acquire Firm 4 after Firm 1 acquires Firm 2 in the good state at date 1 for high enough private benefits.

remaining firm. If no acquisition has yet occurred, Firm 1 acquires Firm 2. Firm 3 has no reason to acquire Firm 4 afterwards, and hence remains passive.

Now we turn to date 0. The last firm to move at that date is Firm 3. If Firm 3 acquires Firm 4, it becomes the second largest firm, since we have assumed $C_3 + C_4 > C_2$. Acquiring Firm 4 makes Firm 3 the most attractive target for Firm 1, which acquires Firm 3 in the good state at date 1 after Firm 3 has acquired Firm 4. Firm 1 pays a price of $C_3 + C_4 + 0.5(C_3 + C_4)$ for the combined firm. Hence, Firm 3's manager receives an expected payoff at date 0 of $w + (1-\rho)(w - \alpha C_4) + \rho \alpha 0.5(C_3 + C_4)$ if Firm 3 acquires Firm 4. If Firm 3 does not acquire Firm 4, Firm 2 remains the second largest firm and hence Firm 1 acquires Firm 2 in the good state at date 1. As a consequence, Firm 3's manager receives a payoff of 2w. Hence, Firm 3 acquires Firm 4 if and only if $w + (1-\rho)(w - \alpha C_4) + \rho \alpha 0.5(C_3 + C_4) + \rho \alpha 0.5(C_3 + C_4) > 2w$ or:

(4) w < 0.5
$$\alpha$$
(C₃ + C₄) - $\alpha \frac{1-\rho}{\rho}$ C₄.

Firm 2 moves before Firm 3 at date 0. If Firm 2 acquires Firm 4 at date 0, it ensures that it is the second largest firm and is acquired by Firm 1 in the good state at date 1. Firm 1 pays a price of $C_2 + C_4 + 0.5(C_2 + C_4)$ for the combined firm. Hence, Firm 2's manager receives an expected payoff at date 0 of up to $w + (1-\rho)(w - \alpha C_4) + \rho \alpha 0.5(C_2 + C_4)$ if Firm 2 acquires Firm 4. If Firm 2 does not acquire Firm 4, two situations can arise. If (4) holds, Firm 3 acquires Firm 4, making it the second largest firm. In this case, Firm 1 acquires the combination of Firms 3 and 4 and Firm 2 remains independent. Hence, Firm 2's manager receives a payoff of 2w. As a consequence, Firm 2 acquires Firm 4 if and only if $w + (1-\rho)(w - \alpha C_4) + \rho \alpha 0.5(C_2 + C_4) > 2w$. One can easily show that this implies that Firm 2 acquires Firm 4 whenever (4) holds.

If (4) does not hold, Firm 3 remains passive and does not acquire Firm 4 if Firm 2 does not acquire Firm 4. In this case, Firm 2 remains the second largest firm and is acquired by Firm 1 for a price of $C_2 + 0.5C_2$ in the good state at date 1. Hence, Firm 2's manager receives an expected payoff at date 0 of $w + (1-\rho)w + \rho\alpha 0.5C_2$. Given this, Firm 2 never acquires Firm 4 if (4) does not hold. Hence, Firm 2 acquires Firm 4 if and only if (4) holds. One can also show that Firm 2 always prefers to acquire Firm 4 to acquiring Firm 3. Proposition 3 summarizes the equilibrium merger dynamics.

Proposition 3: If private benefits are low enough so that (4) holds, there are two profitable acquisitions: Firm 2 acquires Firm 4 at date 0. Then Firm 1 acquires the combination of Firm 2 and

Firm 4 in the good state at date 1. If private benefits are high enough so that (4) does not hold, there is one profitable acquisition: Firm 1 acquires Firm 2 in the good state at date 1.

In contrast to the previous models, the relationship between private benefits and the number of mergers is reversed. In particular, there are two mergers if private benefits are *low* enough and there is only one merger if private benefits are sufficiently high. The early acquisition of Firm 4 by Firm 2 is *not* defensive. Quite to the contrary, Firm 2 acquires Firm 4 in order to ensure that it is the most attractive target – because it is the largest potential target firm – for Firm 1 in the good state at date 1. If it would not acquire Firm 4, Firm 3 would do so and become the second largest firm and hence the most attractive takeover target for Firm 1. The model in this section has no defensive acquisitions because no firm can use an acquisition to become larger than Firm 1. Instead, firms make acquisitions in order to become the second largest firm or remain the second largest firm so that they are the most attractive target for Firm 1 and can earn a takeover premium. Of course, managers engage in these positioning acquisitions to make their firms more attractive targets only if they care sufficiently strongly about firm value and less about the private benefits of control. This explains why they engage in acquisitions if private benefits are low but avoid acquisitions if private benefits are high.

It should be noted that the date 0 acquisition of Firm 4 by Firm 2 - in contrast to the date 0 acquisition in the previous model - is a positive NPV acquisition for Firm 2. Firm 2 engages in this acquisition only if the takeover premium it can earn $(0.5\rho(C_2 + C_4))$ is worth more than the expected loss in firm value that arises due to the negative synergies in the bad state of the world $((1-\rho)C_4)$. This is implied by condition (4), as can be seen after some algebraic manipulation.²⁶

III. A Model of an Industry in which Some but Not All Firms Are of Similar Size

In section I, we presented two models that analyzed homogenous firm size industries and introduced defensive mergers, which are prominent in such industries. This arises because in these models, all firms were of similar size. For example, in section I.C., even if Firm 4 acquired Firm

²⁶ As in the previous model, it is not clear whether date 0 or date 1 acquisitions are more profitable. For example, if Firm 2 acquires Firm 4 at date 0 and then Firm 1 acquires the combination of Firms 2 and 4 at date 1, either of the two acquisitions can be more profitable. The early acquisition (of Firm 4 by Firm 2) can be more profitable if private benefits are fairly high and the good state of the world is sufficiently high, because the premium Firm 2 pays for Firm 4 decreases in private benefits and the probability of the good state. The reason is that this acquisition will lead to a loss of control of Firm 2's manager in the good state at date 1. In contrast, Firm 1's acquisition premium for the combination of Firms 2 and 4 is not a function of private benefits, since Firm 1's manager stays in control with or without an acquisition.

5, it became larger than Firm 1. Hence, all acquisitions (except by Firm 1) had defensive value. In section II, we analyzed heterogeneous firm size industries and introduced positioning mergers, which are prominent in such industries. The reason was that the largest firm was much larger than all other firms. Even if Firm 2 acquired Firm 3, it was still smaller than Firm 1. Hence, no acquisition could have defensive value. However, acquiring another firm made the combined firm more attractive as a target to the largest firm.

In this section, we present a model in which some firms are of similar size but others much smaller than the largest firm. We refer to this type of industry as a mixed firm size industry. The model in this section analyzes the equilibrium merger dynamics in mixed firm size industries. We show that the merger dynamics are yet again different. Both defensive and positioning mergers can occur, depending on the level of private benefits. The mixed firm size industry is arguably the richest (and perhaps most realistic) industry structure and allows us to derive a number of additional results (for example, on the identity of acquirers and targets) and derive others in a cleaner fashion (such as the relationship between acquirer size, deal size, and the profitability of mergers). However, the previous models in sections I and II are not special cases of this model. They derived equilibrium merger dynamics in industry structures that are not covered by the model in this section. Indeed, one of the primary insights arising from our analysis is that industry structure matters for merger dynamics.

At each date, Firm 4 moves first, Firm 3 second, Firm 2 third, and Firm 1 last. We assume that $C_2 + C_5 > C_1$, $C_3 + C_4 > C_1$, $C_3 + C_5 < C_1$, and $C_4 + C_5 > C_2$. Hence, Firm 2 is large enough so that any acquisition makes it larger than Firm 1 and hence immune against any acquisition. Below, we refer to it as a "large" firm. Firm 4 is small enough so that the only acquisition it can make, acquiring Firm 5, does not make it larger than Firm 1 and hence it cannot become immune against being acquired. Below we refer to it as a "small" firm. Firm 3 is large enough that it can make an acquisition (of Firm 4) that makes it larger than Firm 1 and hence immune against being acquired. However, it is also small enough that it can make an acquisition (of Firm 5) that may increase its attractiveness as a target and hence its likelihood of being acquired. Below we refer to it as a "medium" size firm. Hence, in this model we have "large, "medium", and "small" firms. In contrast, in the two models in section I, all firms were "large", and in the model in section II, all firms (except Firm 1) were "small". The remainder of the model is as before. Proposition 4 gives an informal version of the equilibrium merger dynamics that summarizes the key insights. In Appendix A, we give the exact formulation, which shows the equilibrium merger activity in all of the many different parameter regions.

Proposition 4: If private benefits w are low, Firm 4 or Firm 3 acquires Firm 5 at date 0 or in the good state at date 1. Then Firm 1 acquires the combined firm in the good state at date 1. All these acquisitions are profitable. If private benefits w are high, Firm 3 acquires Firm 4 and Firm 2 acquires Firm 5, either at date 0 or in the good state at date 1. Both acquisitions are unprofitable. If private benefits w are very high, Firm 4 acquires Firm 5 at date 0. For high enough private benefits, this acquisition is unprofitable. Then Firm 1 acquires the combined company in the good state at date 1 in a profitable acquisition.

Unlike in the previous models, there are differences in the behavior of firms (beyond that Firm 1, due to its special role, behaves differently from the other firms). The key new results relative to the earlier models concern the effect of acquirer and target size on the profitability and frequency of acquisitions.

Proposition 4 shows that large firms (Firm 2) tend do negative NPV acquisitions, while small firms (Firm 4) tend do to positive NPV acquisitions. Large firms have the greatest incentive to make defensive acquisitions, because they are large enough to become immune against being acquired. However, they have no possibility to engage in profitable, positioning acquisitions since any acquisition makes them too large to be acquired. Small firms are too small to be able to use an acquisition to deter being a target of the largest firm. However, they have an incentive to become larger so as to increase their attractiveness as a target, thereby earning a takeover premium. As can be seen from the detailed formulation of Proposition 4 in Appendix A, Firm 4's acquisition of Firm 5 is typically (in all but one of the many different parameter regions) profitable. There is only one exception: for the highest realization of the private benefits of control, this acquisition can be unprofitable.²⁷ Medium-size firms (Firm 3) can and may have an incentive to do either defensive or positioning acquisitions. Hence, they sometimes make profitable and sometimes unprofitable acquisitions.

The Proposition also shows that larger acquisitions tend to be less profitable than smaller acquisitions. In particular, acquisitions of Firm 4 are always unprofitable while acquisitions of Firm 5 are profitable, with two exceptions (if Firm 2 acquires Firm 5 and if Firm 4 acquires Firm 5 for very high private benefits).²⁸ The intuition is that a large acquisition is more likely to be motivated by defensives motives (since it makes the acquirer much larger and hence more likely to be larger than Firm 1) while a small acquisition has less defensive, but more positioning, value.

²⁷ While Firm 4 is too small to protect its independence for two periods, this acquisition secures its independence for one period, and Firm 4's manager is willing to overpay for Firm 5 for this reason.

²⁸ This can be seen more clearly from the exact formulation of Proposition 4 in the Appendix.

The insight that acquirer size affects the profitability of acquisitions is already suggested by a comparison of Propositions 2 and 3. Proposition 2 showed that in a model with only large acquirers, all acquisitions (except by the largest firm) are negative NPV acquisitions. However, it was not clear that these results were driven by acquirer size alone. In particular, all targets as well as acquirers were "large" in the model in section I.C. that leads to Proposition 2. Hence, the unprofitable acquisitions might be driven by target size instead of acquirer size. In that model, it would be impossible to disentangle the effects of target size and of acquirer size. Proposition 3 showed that in a model with only small acquirers (except Firm 1), these firms made only positive NPV acquisitions. That model cannot say anything about the importance of target size since small firms can only acquire even smaller firms.

In the model of this section, we can distinguish between the effect of acquirer and target size. Large acquirers make unprofitable acquisitions. Clearly, this is not driven by target size, since even the acquisition of small firms is unprofitable. Moreover, the profitability of the acquisitions by medium-size firms depends on the target size, emphasizing the importance of deal size.

Medium-size firms (Firm 3) have the choice between making a defensive acquisition (acquiring Firm 4) and making an acquisition to make them an attractive target for Firm 1 (by acquiring Firm 5). For low private benefits, Firm 3 acquires Firm 5 to become a more attractive target and hence earn a premium from Firm 1. For high private benefits, Firm 3 acquires Firm 4 to defend itself against being acquired by Firm 1. Hence, for a firm with the opportunity to engage in defensive as well as non-defensive acquisitions, the NPV of the acquisition is decreasing in the ratio of the target's to the acquirer's size. In particular, if Firm 3 acquires a small firm (Firm 5), it engages in a positive NPV acquisition. However, if Firm 3 acquires a larger firm (Firm 4), it engages in a negative NPV acquisition. This is because the acquisition of Firm 4 has defensive value while the acquisition of Firm 5 does not.

As a consequence of the effects of acquirer and target size, the combination of acquirer and target size tends to be larger for high private benefits and hence in negative NPV acquisitions: if Firm 2 acquires Firm 5 or Firm 3 acquires Firm 4 - so that the combined company is larger than Firm 1.²⁹ In contrast, for small private benefits, the acquirer tends to be smaller and the sum of acquirer and target size smaller than the size of Firm 1. Proposition 4 shows also that for high private benefits, there is a tendency for larger firms to be the acquirer.³⁰ In particular, Firm 2 makes acquisitions for high private benefits only.

²⁹ Again, the exception is the acquisition of Firm 5 by Firm 4 for very high private benefits.

³⁰ This is true again with the exception of very high private benefits, in which Firm 4 is the acquirer.

Moreover, there are always two mergers, unlike in the previous models. Hence, mixed firm size industries are most likely to generate merger waves. With this industry structure, some firms have defensive merger opportunities while other firms cannot defend themselves. If private benefits are low, firms without defensive merger opportunities (or firms that have the opportunity to engage in defensive as well as other acquisitions) want to acquire other firms to become larger and hence an attractive target for Firm 1. One of them eventually is acquired by Firm 1 after having acquired another firm itself. On the other hand, if private benefits are high, firms with defensive merger opportunities want to acquire other firms to avoid being acquired. This prevents Firm 1 from acquiring any firm. Thus, for both low and high private benefits, two mergers occur. Finally, the Proposition shows that medium-size firms are the most likely to make acquisitions since they can have both defensive and positioning motives. In particular, Firm 3 engages in acquisitions both for low and high private benefits.³¹ A different inclination to make acquisitions for firms of different size, of course, could not be shown in the previous models, since there were only firms (with the exception of Firm 1) that were of very similar size.³²

IV. Discussion of Assumptions and Model Features

In this section, we discuss some of the assumptions and key model features, including implications on how to relate the model to the empirical evidence.

A. Firm Size and the Anticipation of Mergers

In our models, firm size is given by the stand-alone value of the firms. In principle, firm size should be the market (equity) value of the firms. For example, the anticipation of Firm 4's acquisition of Firm 5 in the model in section I.C. should be reflected in the market values of both firms already at date 0. Firm 4's value before the acquisition should be lower than C_4 because it will engage in a negative NPV acquisition. Firm 5's value should be higher than C_5 because it will

³¹ Firm 4 typically makes acquisitions only for low private benefits. But it acquires Firm 5 if private benefits are very high in the exceptional parameter region discussed before.

³² As in the previous models (except the three-firm model), it is not clear whether date 0 acquisitions are more or less profitable than date 1 acquisitions. For example, if Firms 3 or 4 acquires Firm 5 at date 0, this is always less profitable than Firm 1's acquisition of the combined company at date 1. However, if Firm 2 acquires Firm 5 at date 0 and then Firm 3 acquires Firm 4 at date 1, the early acquisition is always more profitable. This again is related to the fact that the latter acquisition pays for the private benefits that Firm 3's manager receives from the acquisition for sure (since it ensures his staying in control) while the earlier acquisition only pays for Firm 2's manager's staying in control in the good state of the world.

earn a premium. We have ignored this issue and assumed for simplicity that the market does not anticipate the acquisitions, perhaps due to some unmodeled uncertainty.

Anticipating mergers could affect the size ordering of firms. Firm 4 may not be able to acquire Firm 5 since its market value (but not its stand-alone value) may be smaller than Firm 5's. We have checked the robustness of our results to these modeling issues for the three-firm and the four-firm model. In particular, we have analyzed a model in which the premium is as in the model we presented, but limited such that the acquirer is not smaller than the target once the merger is fully anticipated (however, the premium has to be still at least zero). Our results do not change once we adjust the assumptions on the stand-alone values of the firms slightly. However, we have not been able to do the same robustness analysis for the two five-firm models, since the models become exceedingly complicated once all mergers are anticipated. In the empirical implications section below, we assume that an (un)profitable acquisition leads to a positive (negative) acquirer announcement return. This is a reasonable interpretation if acquisitions are not perfectly anticipated, which seems to be a realistic assumption.

If one were to take our model to the data and interpret it literally, firm size would be measured by stand-alone values. Stand-alone values could be estimated from market values by subtracting from the market values the value change from takeovers anticipated by the market. The anticipated value change could be estimated by the probability of making an acquisition as well as being acquired for a firm of particular characteristics (for example, its size and the firm size distribution in its industry) multiplied by the average value gain (or loss) in such an acquisition. We do not believe that stand-alone values would be dramatically different from market values or would give rise to very different empirical findings. There appears to be a high degree of new information in most acquisition announcements, as is evident in the high announcement returns for targets and acquirers. An alternative measure of stand-alone size would be book values since they are not affected by anticipated takeover activities. Indeed, some of the literature uses this as proxy for firm size and in general the findings focusing on book values and market values are very similar (see, e.g., Moeller et al. (2004)). In our empirical work below, this is what we use.

B. Only One Acquisition per Firm and the Role of the Largest Firm

We make the simplifying assumption that each firm can only make one acquisition. Consider the three-firm model. If Firm 2 could engage in two acquisitions, it might acquire Firm 1 in the good state at date 1 after acquiring Firm 3 at date 0. Anticipating this, Firm 1 might want to protect its independence by acquiring Firm 3 at date 0. Such an acquisition would also ensure that Firm 2 would be available as a takeover target in the good state at date 1. Hence, allowing multiple acquisitions per firm could create incentives for Firm 1 to engage in date 0 acquisitions (which may or may not be defensive). This would not affect the main insights from the model, but may perhaps make the three-firm model more realistic. However, it would complicate the analysis. We conjecture that allowing only one acquisition per firm is less restrictive in the other models with four and five firms. There is no reason for the largest firm to make a date 0 acquisition in the four-firm model since even without a date 0 acquisition Firm 1 can always acquire another firm in the good state at date 1 and it benefits from mergers between the other firms since it prefers to acquire the largest possible firm. In the five-firm models, Firm 1 might have an incentive to acquire another firm at date 0, but by the time it moves this acquisition opportunity is unlikely to be still available. However, we do need a limit on the speed with which firms can make acquisitions. We believe that in reality transactions costs and also competition for targets make current firm size an important determinant of future firm size and hence current firm size matters for the probability of being taken over.

In a more general (and realistic) model, acquiring a larger firm would be more difficult but not impossible. Then the largest firm might also have to worry about becoming a takeover target. Hence, the largest firm's behavior in our model is so simple only because of the simplifying assumptions that make our model tractable. As a consequence, one should not draw any empirical implications from the behavior of the largest firm in our model. Of course, our model also abstracts from other reasons why such very large firms may engage in negative NPV acquisitions. For example, their managers may be more interested in empire-building or less disciplined by the takeover market precisely because they are more difficult to acquire.

V. Empirical Implications

As discussed above, the theory developed in this paper can explain several stylized facts about mergers. It is also consistent with existing evidence on defensive acquisitions in the U.S. banking industry presented by Louis (2004). He shows that banks that were takeover targets and then engage in acquisitions are less likely to be acquired than targeted banks that do not make an acquisition subsequently. Moreover, the acquirers in such mergers, which he interprets as defensive, pay higher premia and experience worse and significantly negative abnormal returns than acquirers in nondefensive mergers. In this section, we discuss additional empirical implications of our theory.

Our theory predicts a negative correlation between measures of private benefits and acquirer announcement returns as well as a positive correlation between managerial equity ownership and announcement returns (see Propositions 1, 2, and 4). The only exception is that in heterogeneous firm size industries all acquisitions are profitable and hence there is no correlation between private benefits (or managerial equity stakes) and acquirer returns (see Proposition 3). Our general prediction is consistent with Lewellen, Loderer, and Rosenfeld (1985), who find that managerial stock ownership is positively correlated with acquirer announcement returns.

The second factor affecting the defensive motive for an acquisition is firm size. The existing empirical literature is based on examining absolute firm size, rather than relative firm size. In our model "size" refers to relative size – size relative to the largest firm in the industry with profitable acquisition opportunities. We still believe that our model can explain existing size findings to some extent, because size in the sense of our model and absolute size are highly correlated. Using the data introduced in the next section, the correlation is 0.45 (0.51 with the log of total assets). We predict a negative correlation between acquirer returns and acquirer size because large firms are more likely to engage in defensive acquisitions than small firms (see the discussion after Proposition 4). This implication is consistent with the findings in Kahl and Rosen (2002), who show that acquirer returns are decreasing in acquirer size and negative for very large acquirers, and Moeller, Schlingemann, and Stulz (2004), who find that very large acquirers have negative announcement returns and small acquirers have positive announcement returns.³³

We also predict a negative correlation between acquirer returns and target or deal size, because larger acquisitions are more likely to be defensively motivated (see the discussion after Proposition 4). This is consistent with the evidence in Kahl and Valkanov (2004), who find that acquirer returns are decreasing in the size of the merger transaction. As a consequence of the previous observations on acquirer size and target size, we also predict that the sum of acquirer and target size is positively correlated with private benefits and negatively correlated with acquirer returns. Our theory also predicts that for medium-size firms, the ratio of target to acquirer size is negatively correlated with acquirer returns. In contrast, for small acquirers, we conjecture that the ratio of target to acquirer size and the acquirer's abnormal returns are positively correlated. This does not directly follow from our model, since the smallest firm in our model that can make an acquisition (Firm 4) has only one potential target (Firm 5). However, in a more general model with a sixth firm, acquiring Firm 5 should be more profitable for Firm 4 than acquiring Firm 6, since it makes it a more attractive takeover target (and the premium is proportional to the firm's

³³ Moeller et al. (2004) show that the equally weighted acquirer announcement return is 1.1% but the acquiring firm's shareholders lose on average more than \$25 million upon the announcement. This suggests that a subset of large acquirers have negative announcement returns. They also show that small acquirers (defined as firms whose market capitalization is in the lowest 25% of NYSE firms) have strongly positive announcement returns while large acquirers (top 75% of NYSE firms) have announcement returns of about zero. Kahl and Rosen (2002) show that acquirer announcement returns tend to be monotone in acquirer size and, in particular, negative for the largest acquirers.

size). This is consistent with the finding in Moeller et al. (2004) that acquirer announcement returns are decreasing in the size of the target to the acquirer for large acquirers, but the opposite is true for small acquirers.

Some of the implications discussed so far are also consistent with other managerial merger theories. For example, a negative correlation between acquirer returns and acquirer size could arise if one assumes that small firms suffer less from agency problems than large firms. Managers in large firms tend to own a smaller percentage of the firm's shares than those of small firms (Moeller et al. (2004)). Moreover, large firms have larger or less cohesive boards, which tend to be less effective (Yermack (1996)).³⁴ These theories provide plausible explanations for the size effect. We believe that defensive and positioning motives contribute to the size effect. A negative correlation between deal size and acquirer returns is also consistent with a general empirebuilding hypothesis that suggests that larger acquisitions are more likely to be driven by managerial empire-building. It would not be easy to discriminate between our theory and these alternatives by looking at the acquirer and target size findings alone, because they are shared by our theory as well as the alternatives. It would be difficult to tell apart the importance of relative and absolute size, because they are highly correlated. However, these alternative managerial theories have difficulties explaining some of the other empirical findings mentioned above (for example the concentration of mergers in industries undergoing a regime shift). In the following we spell out some additional implications of our theory that may discriminate more easily between our theory and other managerial theories. Medium-size firms are the most likely to engage in acquisitions – they engage both in defensive as well as positioning acquisitions (see the discussion after Proposition 4). Our model also suggests that industry structure affects merger activity. This is tested in the next section.

VI. An Empirical Test

In our theory, the most important factor driving merger activity is the firm size distribution in an industry. We predict that homogenous firm size industries (or firms of similar size within an industry) are likely to be characterized by defensive merger waves (when private benefits are high) and hence have the worst average abnormal returns for acquirers. Heterogeneous firm size industries should have the best average abnormal returns for acquirers, and merger waves should occur for low private benefits. Our theory suggests that mixed firm size industries should be characterized by intermediate average abnormal returns for acquirers.

³⁴ We are grateful to an anonymous referee for suggesting these explanations.

Moreover, we predict that the probability that a firm makes an acquisition should increase with the share of medium-size firms in its industry. We see this by comparing the equilibrium in Section III (Proposition 4) to those in Sections I and II (Propositions 2 and 3, respectively). There are always two acquisitions in mixed firm size industries, whether private benefits are high or low. In the tests below, we look at all industries and hence pool across situations with high and low private benefits. The key difference between the model in Section III and the earlier models is the presence of firms that make both defensive and positioning acquisitions. Which of these two types of acquisitions they make depends on the level of private benefits. These firms, which we call medium size, must be large enough so that an acquisition deters a reasonable fraction of potential suitors, but small enough that the acquisition does not make them too large to be acquired by all their potential suitors. But as Proposition 4 shows, the additional mergers may not come exclusively from medium-size firms.

Note that it is the presence of medium-size firms rather the coexistence of small and large firms that drives the additional merger activity in mixed firm size industries. If there were only small and large firms, small firms might not engage in positioning acquisitions if private benefits were low, because they could not become larger and hence more attractive targets than the large firms. Medium-size firms, however, can become larger than the large firms through a positioning acquisition and hence will make these acquisitions for low private benefits.

Thus, the hypothesis we test is:

(H1) As the proportion of medium-size firms in an industry increases, firms are more likely to make acquisitions, all else equal.

We start with a sample of firms which are in both the CRSP and Compustat databases in any year during the period 1983 - 2003. Acquisition data consist of completed mergers and acquisitions by U.S. firms announced between 1983 and 2004 as given in the Securities Data Corporation (SDC) database.³⁵ We define a merger as an acquisition of equity where one firm purchases at least 50% of another and, after the purchase, the bidder owns at least 90% of the target. Thus, we do not include gradual acquisitions. We also require that the price paid for the target is at least \$1 million.

In our model, all mergers are within-industry transactions. An industry is, in the sense of our model a collection of firms that are likely to acquire each other. Thus, in the empirical tests, we want industries to be somewhat closed, in the sense that there are not too many cross-industry

mergers. This suggests a relatively small number of industries. But, we also want industry definitions to be narrow enough so that firms view others in the industry to be natural merger partners. Otherwise we would also include many firms in our relative firm size definitions that are extremely unlikely to acquire the firm. This suggests a larger number of industries. We divide firms into 48 industries using the classification given in Fama and French (1997), dropping firms grouped as in the miscellaneous industry.³⁶ In our sample, 44.8 percent of mergers are crossindustry. Using a broader definition of an industry, such as French's 5-industry classification means few cross-industry mergers (14.5 percent), but leaves very different types of firms in the same industry (for example, mining companies are in the same industry as hotels). Our final sample, after dropping firms in the miscellaneous industry grouping, has 103,168 firm-year observations. Table 1 gives summary statistics for the sample. All Tables are in Appendix A.

The hypothesis we test, (H1), requires a definition of a medium-size firm. In the spirit of our model, a medium-size firm is one large enough so that some acquisitions can serve as deterrents, but also small enough so that other mergers can make the firm a more attractive merger target. The precise size range depends on the size of the largest potential suitor for a firm. In the model, for simplicity we assume that the largest suitor is the same as the largest firm. While this is without loss of generality in the model, it is not necessarily true in practice. This means that a firm need not be at least 50% of the size of the largest firm to deter all likely acquisitions. Moreover, firms are not restricted to one acquisition, but can grow larger through a few. For these reasons, we chose the medium size range to capture firms that are not the largest in an industry, but still are relatively large compared to the mass of firms. There are enough firms larger than these firms that they are likely to be threatened by takeover from at least some larger firm, but they are big enough that by making a major acquisition, they may be able to deter some potential acquirers.

We define a medium-size firm as one with total assets that are between 5% and 30% of the total assets of the largest firm in the industry in that year (henceforth referred to as the size ratio). As seen in Table 1, only 2.6% of firms are large when we use a 30% size ratio cutoff between medium and large.³⁷ Since not all of these larger firms have profitable merger opportunities, a medium-size firm may be able to use an acquisition to deter some potential suitors. With a 5%

³⁵ We do not include mergers where the target is listed as a government organization or joint venture.

³⁶ The industry classifications for the 48-industry grouping and the 5-industry grouping we use as a robustness test are available at mba.tuck.dartmouth.edu/pages/faculty/ken.french/data_library.html. We thank Ken French for making these data available. Using 2-digit SIC codes to define industries gives less explanatory power, but similar results.

³⁷ In roughly 20% of the industry-year observations, there is no other firm that is more than 50% of the size of the largest firm in the industry.

size ratio cutoff between small and medium, we classify roughly 8% of firms as medium-size. This means that medium-size firms are, as we desire, large relative to the mass of smaller firms.

Let *Future merger*_{*j*,*i*,*t*} be a dummy variable that is one if and only if firm j in industry i announces an acquisition in year t+1 and the acquisition is subsequently completed (that is, we classify mergers by their announcement date). The mean value of Future Merger is 0.118 which means that there was at least one acquisition in 12,154 firm-year observations. The total number of mergers in our sample is larger than this, since many firms made more than one acquisition in a calendar year. The sample contains 19,699 mergers with a total deal value of \$5.2 trillion.

To test whether the distribution of firm sizes affects acquisition activity, we use the regression:

(5) *Future merger*_{*j*,*i*,*t*} = f(*Percent of medium-size firms*_{*i*,*t*}, firm-specific controls, year dummies, industry dummies),

where *Percent of medium-size firms*_{*i*,*t*} is the proportion of firms in industry i in year t that are between 5% and 30% of the asset size of the largest firm in the industry. In all the regressions we calculate robust standard errors, including controls for firm-level cluster effects. Hypothesis (H1) predicts that the coefficient on *Percent of medium-size firms*_{*i*,*t*} is positive.

We use firm-specific controls that other studies suggest may affect merger activity. There is evidence that (absolute) firm size can affect the probability of making an acquisition (e.g., Harford (1999)), so we include the log of total assets as a control (Log assets_{it}). We also include a squared log assets term to control for any non-linearity in the relationship between size and merger activity. Firm health is accounted for using the increase in a firm's stock price in year t (Stock market return_i) and the firm's market-to-book ratio (Market/book_i). Strong performance can give firms more currency to make acquisitions or it may indicate that the firm is overvalued, making an acquisition more likely (Harford (2005), Golbe and White (1988), and Shleifer and Vishny (2003)). Q-theory suggests that a firm's investment rate (possibly including mergers) should be increasing in the market-to-book ratio (Jovanovic and Rousseau (2002)). Capital structure can also play a role in merger decisions since firms with high leverage have an incentive to underinvest (Myers (1977)). This leads us to follow other studies by including Equity/asset_{i,t}, the (accounting) leverage ratio, as a control. Firms with more free cash flow may be more able or more willing to make acquisitions. We control for this using the ratio of EBITDA to sales (Andrade and Stafford (2004)). Some firms may also be more likely than others to engage in acquisitions, possibly as part of a growth program (Rosen (2005)). Thus, we include the growth

rate of assets in year t (*Asset growth rate*_{*j*,*t*}) and a dummy for whether the firm made an acquisition in year t (*Recent acquisition*_{*j*,*t*}) as controls.³⁸ We include the industry's Herfindahl index (*Industry Herfindahl*) because industry concentration could affect merger activity. The year dummies pick up any economy-wide factors that affect the probability of merging and the industry dummies capture industry fixed effects.

The first column of Table 2 presents our baseline regression results. The coefficient on *Percent of medium-size firms* is positive and significant at the 1% level indicating that firms in industries with a larger proportion of medium-size firms are more likely to make acquisitions, consistent with (H1). This new result provides evidence in support of our theoretical model. The effect is fairly large. One can calculate from regression specification (1) that a one standard deviation increase in the percentage of medium-size firms in its industry increases the probability of a firm making an acquisition by 2.2 percentage points. This is an increase of about 18.2% in the mean acquisition probability.

The other coefficients generally are consistent with intuition. Larger firms make more acquisitions (although the squared term enters negatively) as do firms with prior acquisitions, better recent performance, faster recent growth, and lower leverage. The coefficient on the industry Herfindahl index is insignificant. This may be because we include industry dummies and there is little variation in an industry's Herfindahl index over time.

In the model, mergers affect the likelihood that a firm is acquired because they change the firm's size. This suggests that acquisitions of very small targets may have little impact on future acquisition activity. For this reason, as a robustness check, we run our tests using different thresholds for what constitutes a "significant" merger. We let *Ratio* be the ratio of the value offered for the target to the acquirer's market value.³⁹ We present results where qualifying mergers have a minimum *Ratio* of 5%, 10%, and 25%. As shown in Table 1, increasing the cutoff significantly reduces the number of mergers that we include as qualifying acquisitions. With no cutoff, 11.8% percent of firms made an acquisition to 7.5%. At a 25% cutoff, only 2.9% of firms make a qualifying acquisition.

The final three columns of Table 2 present regression results for the different minimum ratios of deal value to acquirer size for mergers. The coefficient on *Percent of medium-size firms* is

³⁸ With the exception of asset size and the recent acquisition dummy, we winsorize all the firm-specific controls at the 1% level.

³⁹ We use the value offered for the target rather than the value of the target because many targets are not publicly traded, and thus we do not have a value for them.

positive and significant at least at the 5% level in all the regressions. This provides additional support for (H1).

We do several tests to illustrate the robustness of our results by examining an alternative industry definition and changes in our size ratio cutoffs. The first column of Table 3 give regression results for the baseline model when firms are grouped into 5 Fama-French industries.⁴⁰ The coefficient on *Percent of medium-size firms* is positive and significant, but only at the 10% level. Moreover, the magnitude of the effect is much smaller than in the specification with 48 Fama-French industries, as can be calculated from the regression. This is reassuring, since one would expect the firm size distribution to matter less if industries are very broadly defined and hence firms are not as natural merger partners. We also run our baseline model for different definitions of a medium-size firm. The second column of Table 3 reports results when medium-size firms are those with between 5% and 40% of the assets of the largest firm in the industry and the third column of Table 3 reports results when medium-size firms are those with between 10% and 60% of the assets of the largest firm in the industry. As indicated by the results, we maintain statistical significance (although only at the 10% level for the last definition) on *Percent of medium-size firms* when we change the definition of a medium-size firm. In general, our robustness tests indicate support for (H1).

VII. Conclusion

In this paper we have argued that in industries with economies of scale, firm size is driving mergers. Firms may engage in a race for firm size for two reasons. First, managers may want to make acquisitions to increase their firm's size and hence reduce the likelihood that it is taken over. These acquisitions may allow them to preserve their private benefits of control but are unprofitable. Second, firms may want to engage in profitable acquisitions to become larger and hence position themselves as more attractive takeover targets. We show that industry structure – the size distribution of the firms in the industry – is a very important determinant of which mergers occur. Industries in which many firms are similar in size to the largest firm with profitable acquisition opportunities are prone to defensive merger waves if managers care a lot about the private benefits of control. However, industries in which firms are of very different size are less prone to waves of unprofitable acquisitions. In contrast, they may display merger waves of profitable acquisitions if managers care little about private benefits of control. In industries in which some but not all firms are of similar size, merger waves are most likely, because they occur if private benefits are either low or high. The profitability of acquisitions tends to decrease in the

acquirer's size. Large acquirers overpay while small acquirers tend to engage in profitable acquisitions. Firms of intermediate size sometimes engage in profitable and sometimes in unprofitable acquisitions. Overall, the race for firm size often leads to profitable acquisitions, but if private benefits are high, it may induce large firms to make unprofitable acquisitions.

Our theory can explain why mergers are concentrated in industries for which a regime shift can be identified. At the same time, it can explain why (at least very large) acquirers lose money, on average, and why mergers often occur in waves. The theory generates many additional testable empirical implications. Some of these predictions are consistent with the available empirical evidence, such as the negative correlation between acquirer announcement returns and both acquirer size (Kahl and Rosen (2002) and Moeller, Schlingemann, and Stulz (2004)) and the size of the merger transaction (Kahl and Valkanov (2004)) as well as the worse acquirer returns in defensive acquisitions (Louis (2004)). We caution, though, that the notion of size in our model is slightly different than, although highly correlated with, absolute size as used in the empirical literature. We test one empirical implication that is central to our theory, namely, that industry firm size distribution matters for merger activity. In particular, we test whether, as predicted by our model, firms in industries with more medium-size firms are more likely to make acquisitions. Our evidence supports this prediction. We leave the testing of our other predictions to future empirical work.

⁴⁰ As noted earlier, this classification scheme comes from French's web site.

Appendix A: Tables

Table 1. Summary statistics

The table contains summary statistics for the 103,168 firm-year observations. For observations in year t, Future merger takes the value 1 if a firm acquires another firm in year t + 1, and is otherwise 0. Recent acquisition takes the value 1 if a firm acquires another firm in year t. For both merger variables, the ratio cutoffs are based on the ratio of the value paid for the target divided by the market value of the acquirer at the end of the year prior to a merger announcement. An X% cutoff includes all mergers with a ratio above X%. With the exception of the stock market return and asset growth rate, all other variables are as of the end of year t. The size ratio is the ratio of the total assets of the firm divided by the total assets of the largest firm in its industry. There are 48 industries as defined by Fama and French (1997), but we drop firms in the miscellaneous group. The stock market return is the return for a firm's stock in year t. Market/book is the ratio of the market value of equity to the book value of equity. Equity/assets is the ratio of book equity to total assets. EBITDA/sales is the ratio of earnings before interest, taxes, depreciation, and amortization divided by total sales. Asset growth rate is the rate of growth of total assets in year t.

	Mean	Median	Standard de- viation
Future merger	0.118	0	0.322
Future merger (5% ratio cutoff)	0.075	0	0.263
Future merger (10% ratio cutoff)	0.056	0	0.229
Future merger (25% ratio cutoff)	0.029	0	0.167
Share of industry that is small (size ratio < 5%)	0.894	0.941	0.113
Share of industry that is medium size (size ratio $\in [5\%, 30\%]$)	0.080	0.042	0.081
Share of industry that is large (size ratio > 30%)	0.026	0.013	0.035
Total assets (\$ million)	3,495	182	23,835
Stock market return	0.166	0.067	0.654
Market/book	2.533	1.495	3.530
Equity/asset	0.468	0.460	0.245
EBITDA/sales	-0.041	0.113	1.091
Asset growth rate	0.137	0.048	0.422
Industry Herfindahl	0.113	0.077	0.098
Recent acquisition	0.130	0	0.336
Recent acquisition (5% ratio cutoff)	0.090	0	0.286
Recent acquisition (10% ratio cutoff)	0.067	0	0.251
Recent acquisition (25% ratio cutoff)	0.035	0	0.185

Table 2. Regressions of the probability of being an acquirer on percent of medium-size firms.

The sample contains 103,168 firm-year observations. The dependent variable is Future merger. For observations in year t, Future merger takes the value 1 if a firm acquires another firm in year t + 1, and is otherwise 0. Recent acquisition takes the value 1 if a firm acquires another firm in year t. For both merger variables, the ratio cutoffs are based on the ratio of the value paid for the target divided by the market value of the acquirer at the end of the year prior to a merger announcement. An X% cutoff includes all mergers with a ratio above X%. With the exception of the stock market return and asset growth rate, all other variables are as of the end of year t. Log assets is the natural log of total assets. The percent of industry that is medium size is the percent of a firm's industry where the size ratio is between 5% and 30%, where the size ratio is the ratio of the total assets of the firm divided by the total assets of the largest firm in its industry. There are 48 industries (Fama and French, 1997), but we drop firms in the miscellaneous group. The stock market return is the return for a firm's stock in year t. Market/book is the ratio of the market value of equity to the book value of equity. Equity/assets is the ratio of book equity to total assets. EBITDA/sales is the ratio of earnings before interest, taxes, depreciation, and amortization divided by total sales. Asset growth rate is the rate of growth of total assets in year t. The industry Herfindahl is the Herfindahl index for the firm's industry.

	(1)	(2)	(3)	(4)
	All mergers	Ratio > 5%	<i>Ratio</i> > 10%	<i>Ratio</i> > 25%
Pct of industry that is	0.642	0.863	0.792	1.067
medium size	(0.008)***	(0.004)***	(0.020)**	(0.022)**
Log assets	0.457	0.398	0.311	0.189
	(0.000)***	(0.000)***	(0.000)***	(0.000)***
$(Log assets)^2$	-0.017	-0.022	-0.019	-0.014
	(0.000)***	(0.000)***	(0.000)***	(0.001)***
Stock market return	0.228	0.219	0.219	0.164
	(0.000)***	(0.000)***	(0.000)***	(0.000)***
Market/book	0.039	0.012	0.001	-0.009
	(0.000)***	(0.005)***	(0.909)	(0.177)
Equity/asset	0.874	0.364	0.105	-0.297
	(0.000)***	(0.000)***	(0.155)	(0.002)***
EBITDA/sales	0.082	0.061	0.052	0.032
	(0.000)***	(0.001)***	(0.010)***	(0.170)
Asset growth rate	0.190	0.272	0.270	0.314
_	(0.000)***	(0.000)***	(0.000)***	(0.000)***
Industry Herfindahl	-0.205	-0.115	-0.315	-0.075
-	(0.323)	(0.663)	(0.310)	(0.852)
Recent acquisitions	1.161	1.011	0.927	0.892
-	(0.000)***	(0.000)***	(0.000)***	(0.000)***

Year and industry dummies not shown.

Recent acquisitions includes only mergers meeting ratio restrictions.

Robust p values in parentheses

* significant at 10%; ** significant at 5%; *** significant at 1%

Table 3. Robustness regressions of the probability of being an acquirer on percent of medium-size firms.

The dependent variable is Future merger. For observations in year t, Future merger takes the value 1 if a firm acquires another firm in year t + 1, and is otherwise 0. Recent acquisition takes the value 1 if a firm acquires another firm in year t. For both merger variables, the ratio cutoffs are based on the ratio of the value paid for the target divided by the market value of the acquirer at the end of the year prior to a merger announcement. An X% cutoff includes all mergers with a ratio above X%. With the exception of the stock market return and asset growth rate, all other variables are as of the end of year t. Log assets is the natural log of total assets. In the regressions in the first column, the percent of industry that is medium size is the percent of a firm's industry where the size ratio is between 5% and 30%, where the size ratio is the ratio of the total assets of the firm divided by the total assets of the largest firm in its industry. The regression in the second column defines a medium-size firm as one with a size ratio between 5% and 40% while for the regression in the last column, a medium-size firm is one with a size ratio between 10% and 60%. The regression in the first column uses 5 industries as defined by French and the last two columns use 48 industries as defined by Fama and French (1997) (dropping firms in the miscellaneous group). The stock market return is the return for a firm's stock in year t. Market/book is the ratio of the market value of equity to the book value of equity. Equity/assets is the ratio of book equity to total assets. EBITDA/sales is the ratio of earnings before interest, taxes, depreciation, and amortization divided by total sales. Asset growth rate is the rate of growth of total assets in year t. The industry Herfindahl is the Herfindahl index for the firm's industry.

	(1)	(2)	(3)
	5 industry groups	Medium size: size	Medium size: size
		ratio ∈ [5%, 40%]	ratio ∈ [10%,60%]
	1 000	0.5.0	0.500
Pct of industry that is	1.008	0.569	0.520
medium size	(0.085)*	(0.012)**	(0.067)*
Log assets	0.424	0.457	0.457
C	(0.000)***	(0.000)***	(0.000)***
$(Log assets)^2$	-0.016	-0.017	-0.017
	(0.000)***	(0.000)***	(0.000)***
Stock market return	0.228	0.228	0.228
	(0.000)***	(0.000)***	(0.000)***
Market/book	0.037	0.039	0.039
	(0.000)***	(0.000)***	(0.000)***
Equity/asset	0.698	0.874	0.876
	(0.000)***	(0.000)***	(0.000)***
EBITDA/sales	0.109	0.082	0.082
	(0.000)***	(0.000)***	(0.000)***
Asset growth rate	0.195	0.190	0.190
-	(0.000)***	(0.000)***	(0.000)***
Industry Herfindahl	-0.721	-0.205	-0.268
	(0.103)	(0.326)	(0.200)
Recent acquisitions	1.212	1.161	1.161
_	(0.000)***	(0.000)***	(0.000)***
Observations	103,713	103,168	103,168

Year and industry dummies not shown.

Robust p values in parentheses

* significant at 10%; ** significant at 5%; *** significant at 1%

References

Agrawal, Anup, and Gershon N. Mandelker (1987), "Managerial Incentives and Corporate Investment and Financing Decisions," *Journal of Finance* 42(4): 823-837.

Ambrose, Brent W., and William L. Megginson (1992), "The Role of Asset Structure, Ownership Structure, and Takeover Defenses in Determining Acquisition Likelihood," *Journal of Financial and Quantitative Analysis* 27(4): 575-589.

Amihud, Yakov and Baruch Lev (1981), "Risk Reduction as Managerial Motive for Conglomerate Mergers," *Bell Journal of Economics* 12(2): 823-837.

Andrade, Gregor, Mark Mitchell, and Erik Stafford (2001), "New Evidence and Perspectives on Mergers," *Journal of Economic Perspectives* 15(2): 103-120.

Ang, James S., and Yingmei Cheng (2003), "Direct Evidence on the Market-Driven Acquisitions Theory," working paper, Florida State University.

Andrade, Gregor, and Erik Stafford (2004), "Investigating the Economic Role of Mergers," *Journal of Corporate Finance* 10(1): 1-36.

Asquith, Paul, Robert Bruner, and David Mullins (1987), "Merger Returns and the Form of Financing," mimeo.

Baker, J. (1997), "Contemporary Merger Analysis," George Mason Law Review 5(3): 347-361.

Banerjee, Ajeyo, and James Owers (1992), "Wealth Reduction in White Knight Bids," *Financial Management* 21(3): 48-57.

Baumol, William (1959), <u>Business Behavior, Value, and Growth</u>, New York, New York: Macmillan.

Bebchuk, Lucian, John Coates IV, and Guhan Subramanian (2002), "The Powerful Antitakeover Force of Staggered Boards: Theory, Evidence, and Policy," *Stanford Law Review* 54: 887-951.

Berry, Steven and Ariel Pakes (1993), "Some Applications and Limitations of Recent Advances in Empirical I.O.: Merger Analysis," *American Economic Review* 83: 247-252.

Bliss, Richard, and Richard Rosen (2001), "CEO Compensation and Bank Mergers," *Journal of Financial Economics* 61(1): 107-138.

Bradley, Michael, Anand Desai, and E. Han Kim (1988), "Synergistic Gains from Corporate Acquisitions And Their Division Between the Stockholders Of Target and Acquiring Firms," *Journal of Financial Economics* 21 (1): 3-40.

Bradley, Michael, and Anant Sundaram (2004), "Do Acquisitions Drive Performance or Does Performance Drive Acquisitions?," working paper, Duke University.

Byrd, John, and Kent Hickman (1992), "Do Outside Directors Monitor Managers?: Evidence From Tender Offer Bids," *Journal of Financial Economics* 32 (2): 195-221.

Comment, Robert, and G. William Schwert (1995), "Poison or Placebo? Evidence on the Deterrence and Wealth Effects of Modern Antitakeover Measures," *Journal of Financial Economics* 39 (1): 3-43.

Dong, Ming, David Hirshleifer, Scott Richardson, and Siew Hong Teoh (2003), "Does Investor Misvaluation Drive the Takeover Market?," working paper, Ohio State University.

Epstein, Roy and Daniel Rubinfeld (2001), "Merger Simulation: A Simplified Approach with New Applications," University of California, Berkeley, Working Paper #E01-309.

Ericson, Richard and Ariel Pakes (1995), "An Alternative Theory of Firm and Industry Dynamics," *Review of Economic Studies* 62: 53-82.

Fama, Eugene and Kenneth French, 1997, "Industry Costs of Equity," *Journal of Financial Economics* 43(2): 153-193.

Gilson, Ronald J., and Bernard S. Black (1995), <u>The Law And Finance Of Corporate Acquisitions</u>, Westbury, New York: The Foundation Press, 2nd edition.

Gilson, Stuart C. (1989), "Management Turnover and Financial Distress," *Journal of Financial Economics* 25(2): 241-262.

Golbe, D.L. and L.J. White (1988), "A Time Series Analysis of Mergers and Acquisitions in the U.S. Economy." In: Auerbach, A. (Ed.), <u>Corporate Takeovers: Causes and Consequences</u>. NBER: University of Chicago Press, Chicago, pp. 265-302.

Gort, Michael (1969), "An Economic Disturbance Theory of Mergers," *Quarterly Journal of Economics* 83(4): 624-642.

Gowrisankaran, Gautam (1999), "A Dynamic Model of Endogenous Horizontal Mergers," *Rand Journal of Economics* 30: 56-83.

Gowrisankaran, Gautam and Thomas J. Holmes (2004), "Mergers and the Evolution of Industry Concentration: Results from the Dominant-Firm Model," *RAND Journal of Economics* 35(3): 561-582.

Harford, Jarrad (1999), "Corporate Cash Reserves and Acquisitions," *Journal of Finance* 54(6): 1969-1997.

Harford, Jarrad (2005), "What Drives Merger Waves?," *Journal of Financial Economics* 77: 529-560.

Harris, Ellie (1994), "Why One Firm Is The Target and The Other The Bidder In Single-Bidder, Synergistic Takeovers," *Journal of Business* 67 (2): 263-280.

Hart, Oliver, and John Moore (1995), "Debt and Seniority: An Analysis of the Role of Hard Claims in Constraining Management," *American Economic Review* 85(3): 567-585.

Hasbrouck, Joel (1985), "The Characteristics of Takeover Targets: Q and Other Measures," *Journal of Banking and Finance* 9(3): 351-362.

Jarrell, Gregg, James Brickley, and Jeffrey Netter (1988), "The Market for Corporate Control: The Empirical Evidence since 1980," *Journal of Economic Perspectives* 2(1): 49-68.

Jennings, Robert, and Michael Mazzeo (1991), "Stock Price Movements around Acquisition Announcements and Management's Response," *Journal of Business* 64(2): 139-163.

Jensen, Michael, and Richard Ruback (1983), "The Market for Corporate Control: The Scientific Evidence," *Journal of Financial Economics* 11(1-4): 5-50.

Jensen, Michael C. (1986), "Agency Costs of Free Cash Flow, Corporate Finance, and Takeovers," *American Economic Review* 76(2): 323-329.

Jensen, Michael C. (1993), "The Modern Industrial Revolution, Exit, and the Failure of Internal Control Systems," *Journal of Finance* 48(3): 831-880.

Jovanovic, Boyan, and Serguey Braguinsky (2004), "Bidder Discounts and Target Premia in Takeovers," *American Economic Review* 94(1): 46-56.

Jovanovic, Boyan, and Peter Rousseau (2002), "The Q-Theory of Mergers," *American Economic Review* 92(2): 198-204.

Jovanovic, Boyan and Peter L. Rousseau (2002), "Mergers and Technological Change: 1885-1998," working paper.

Kahl, Matthias, and Richard Rosen (2002), "Acquirer Size, Abnormal Returns, and The Managerial View of Mergers," working paper, University of California, Los Angeles.

Kahl, Matthias, and Rossen Valkanov (2004), "Mega-Mergers and Stock Market Booms," working paper, University of North Carolina at Chapel Hill.

Kamien, Morton and Israel Zang (1990), "The Limits of Monopolization through Acquisition," *Quarterly Journal of Economics* 105: 465-499.

Lambrecht, Bart M. (2004), "The Timing and Terms of Mergers Motivated by Economies of Scale," *Journal of Financial Economics* 72(1): 41-62.

Lewellen, Wilbur, Claudio Loderer, and Ahron Rosenfeld (1985), "Merger Decisions and Executive Stock Ownership in Acquiring Firms," *Journal of Accounting and Economics* 7 (1-3): 209-231.

Louis, Henock (2004), "The Cost of Using Bank Mergers as Defensive Mechanisms against Takeover Threats," *The Journal of Business* 77(2): 295-310.

Malmendier, Ulrike, and Geoffrey Tate (2003), "Who Makes Acquisitions? CEO Overconfidence and the Market's Reaction," working paper, Stanford University.

Marris, Robin (1964), <u>The Economic Theory of Managerial Capitalism</u>, Glencoe, Illinois: Free Press.

May, Don (1995), "Do Managerial Motives Influence Firm Risk Reduction Strategies?," *Journal of Finance* 50(4): 1291-1308.

McCardle, Kevin F., and S. Viswanathan (1994), "The Direct Entry versus Takeover Decision and Stock Price Performance around Takeovers," *Journal of Business* 67(1): 1-43.

Mitchell, Mark L., and J. Harold Mulherin (1996), "The Impact of Industry Shocks on Takeover and Restructuring Activity," *Journal of Financial Economics* 41(2): 193-229.

Moeller, Sara B., Frederik P. Schlingemann, and Rene M. Stulz (2004), "Firm Size and the Gains from Acquisitions," *Journal of Financial Economics* 73(2): 201-228.

Morck, Randall, Andrei Shleifer, and Robert W. Vishny (1988), "Characteristics of Targets of Hostile and Friendly Takeovers," in: Alan J. Auerbach, editor, <u>Corporate Takeovers: Causes and Consequences</u>, Chicago, Illinois: University of Chicago Press: 101-129.

Morck, Randall, Andrei Shleifer, and Robert Vishny (1990), "Do Managerial Objectives Drive Bad Acquisitions?," *Journal of Finance* 45(1): 31-48.

Morellec, Erwan, and Alexei Zhdanov (2005), "The Dynamics of Mergers and Acquisitions," *Journal of Financial Economics* 77(3): 649-672.

Myers, Stewart C. (1977), "Determinants of Corporate Borrowing," *Journal of Financial Economics* 5(2): 147-175.

Palepu, Krishna G. (1986), "Predicting Takeover Targets: A Methodological and Empirical Analysis," *Journal of Accounting and Economics* 8(1): 3-35.

Peters, Craig (2003), "Evaluating the Performance of Merger Simulation: Evidence from the U.S. Airline Industry," Center for the Study of Industrial Organization, Northwestern University, Working Paper #0032.

Rau, Raghavendra P., and Theo Vermaelen (1998), "Glamour, Value, and the Post-acquisition Performance of Acquiring Firms," *Journal of Financial Economics* 49(2): 223-253.

Rhodes-Kropf, Matthew, David T. Robinson, and S. Viswanathan (2005), "Valuation Waves and Merger Activity: The Empirical Evidence," *Journal of Financial Economics* 77(3): 561-603.

Rhodes-Kropf, Matthew, and S. Viswanathan (2004), "Market Valuation and Merger Waves," *Journal of Finance* 59(6): 2685-2718.

Roll, Richard (1986), "The Hubris Hypothesis Of Corporate Takeovers," *Journal of Business* 59(2): 197-216.

Rosen, Richard J. (2005), "Betcha Can't Acquire Just One: Merger Programs and Compensation," working paper.

Ruback, Richard (1988), "An Overview of Takeover Defenses," in Alan J. Auerbach, editor, <u>Mergers and Acquisitions</u>, University of Chicago Press, Chicago: 69-85.

Rubin, Paul H. (1973), "The Expansion of Firms," Journal of Political Economy 81(4): 936-949.

Schwert, G. William (2000), "Hostility in Takeovers: In the Eyes of the Beholder," *Journal of Finance*, 55: 2599-2640.

Servaes, Henri (1991), "Tobin's Q and the Gains From Takeovers," *Journal of Finance* 46(1): 409-419.

Shleifer, Andrei, and Robert W. Vishny (1989), "Management Entrenchment: The Case of Manager-Specific Investments," *Journal of Financial Economics* 25(1): 123-139.

Shleifer, Andrei, and Robert W. Vishny (2003), "Stock Market Driven Acquisitions," *Journal of Financial Economics* 70(3): 295-311.

Toxvaerd, Flavio (2004), "Strategic Merger Waves: A Theory of Musical Chairs," working paper, Hebrew University of Jerusalem.

Varaiya, Nikhil, and Kenneth Ferris, 1987, "Overpaying in Corporate Takeovers: The Winner's Curse," *Financial Analysts' Journal*: 64-70.

Weston, J. Fred, Kwang S. Chung, and Juan A. Siu (1998), <u>Takeovers, Restructuring, and Corporate Governance</u>, Upper Saddle River, NJ, Prentice Hall.

Yermack, David (1996), "Higher Market Valuation of Companies with a Small Board of Directors," *Journal of Financial Economics* 40(2):185-211.

You, Victor, Richard Caves, Michael Smith, and James Henry (1986), "Mergers and Bidders' Wealth: Managerial And Strategic Factors," in: Lacy Glenn Thomas, III (ed.), <u>The Economics of Strategic Planning. Essays In Honor Of Joel Dean</u>, Lexington Books, 201-221.

Figure 1: Timing in the Three-firm Model

Date 0	
	Firms learn parameters of the model.
	Firm 1 decides whether to make an acquisition offer, and, if
	so, which Firm to acquire.
	Target firm shareholders decide whether to accept the acquisition offer.
	Firm 2, if it has not been acquired, decides whether to make
	an acquisition offer, and, if so, which Firm to acquire.
	Target firm shareholders decide whether to accept the acquisition offer.
Date 1	
	Firms learn state of nature: either good or bad.
Ţ	In both states, repeat acquisition moves as at date 0.

Working Paper Series

A series of research studies on regional economic issues relating to the Seventh Federal Reserve District, and on financial and economic topics.

A Proposal for Efficiently Resolving Out-of-the-Money Swap Positions at Large Insolvent Banks George G. Kaufman	WP-03-01
Depositor Liquidity and Loss-Sharing in Bank Failure Resolutions George G. Kaufman	WP-03-02
Subordinated Debt and Prompt Corrective Regulatory Action Douglas D. Evanoff and Larry D. Wall	WP-03-03
When is Inter-Transaction Time Informative? Craig Furfine	WP-03-04
Tenure Choice with Location Selection: The Case of Hispanic Neighborhoods in Chicago Maude Toussaint-Comeau and Sherrie L.W. Rhine	WP-03-05
Distinguishing Limited Commitment from Moral Hazard in Models of Growth with Inequality* Anna L. Paulson and Robert Townsend	WP-03-06
Resolving Large Complex Financial Organizations Robert R. Bliss	WP-03-07
The Case of the Missing Productivity Growth: Or, Does information technology explain why productivity accelerated in the United States but not the United Kingdom? Susanto Basu, John G. Fernald, Nicholas Oulton and Sylaja Srinivasan	WP-03-08
Inside-Outside Money Competition Ramon Marimon, Juan Pablo Nicolini and Pedro Teles	WP-03-09
The Importance of Check-Cashing Businesses to the Unbanked: Racial/Ethnic Differences William H. Greene, Sherrie L.W. Rhine and Maude Toussaint-Comeau	WP-03-10
A Firm's First Year Jaap H. Abbring and Jeffrey R. Campbell	WP-03-11
Market Size Matters Jeffrey R. Campbell and Hugo A. Hopenhayn	WP-03-12
The Cost of Business Cycles under Endogenous Growth Gadi Barlevy	WP-03-13
The Past, Present, and Probable Future for Community Banks Robert DeYoung, William C. Hunter and Gregory F. Udell	WP-03-14

Measuring Productivity Growth in Asia: Do Market Imperfections Matter? John Fernald and Brent Neiman	WP-03-15
Revised Estimates of Intergenerational Income Mobility in the United States Bhashkar Mazumder	WP-03-16
Product Market Evidence on the Employment Effects of the Minimum Wage Daniel Aaronson and Eric French	WP-03-17
Estimating Models of On-the-Job Search using Record Statistics Gadi Barlevy	WP-03-18
Banking Market Conditions and Deposit Interest Rates <i>Richard J. Rosen</i>	WP-03-19
Creating a National State Rainy Day Fund: A Modest Proposal to Improve Future State Fiscal Performance <i>Richard Mattoon</i>	WP-03-20
Managerial Incentive and Financial Contagion Sujit Chakravorti and Subir Lall	WP-03-21
Women and the Phillips Curve: Do Women's and Men's Labor Market Outcomes Differentially Affect Real Wage Growth and Inflation? Katharine Anderson, Lisa Barrow and Kristin F. Butcher	WP-03-22
Evaluating the Calvo Model of Sticky Prices Martin Eichenbaum and Jonas D.M. Fisher	WP-03-23
The Growing Importance of Family and Community: An Analysis of Changes in the Sibling Correlation in Earnings <i>Bhashkar Mazumder and David I. Levine</i>	WP-03-24
Should We Teach Old Dogs New Tricks? The Impact of Community College Retraining on Older Displaced Workers <i>Louis Jacobson, Robert J. LaLonde and Daniel Sullivan</i>	WP-03-25
Trade Deflection and Trade Depression Chad P. Brown and Meredith A. Crowley	WP-03-26
China and Emerging Asia: Comrades or Competitors? Alan G. Ahearne, John G. Fernald, Prakash Loungani and John W. Schindler	WP-03-27
International Business Cycles Under Fixed and Flexible Exchange Rate Regimes <i>Michael A. Kouparitsas</i>	WP-03-28
Firing Costs and Business Cycle Fluctuations Marcelo Veracierto	WP-03-29
Spatial Organization of Firms Yukako Ono	WP-03-30
Government Equity and Money: John Law's System in 1720 France François R. Velde	WP-03-31

Deregulation and the Relationship Between Bank CEO Compensation and Risk-Taking Elijah Brewer III, William Curt Hunter and William E. Jackson III	WP-03-32
Compatibility and Pricing with Indirect Network Effects: Evidence from ATMs Christopher R. Knittel and Victor Stango	WP-03-33
Self-Employment as an Alternative to Unemployment <i>Ellen R. Rissman</i>	WP-03-34
Where the Headquarters are – Evidence from Large Public Companies 1990-2000 <i>Tyler Diacon and Thomas H. Klier</i>	WP-03-35
Standing Facilities and Interbank Borrowing: Evidence from the Federal Reserve's New Discount Window <i>Craig Furfine</i>	WP-04-01
Netting, Financial Contracts, and Banks: The Economic Implications William J. Bergman, Robert R. Bliss, Christian A. Johnson and George G. Kaufman	WP-04-02
Real Effects of Bank Competition Nicola Cetorelli	WP-04-03
Finance as a Barrier To Entry: Bank Competition and Industry Structure in Local U.S. Markets? <i>Nicola Cetorelli and Philip E. Strahan</i>	WP-04-04
The Dynamics of Work and Debt Jeffrey R. Campbell and Zvi Hercowitz	WP-04-05
Fiscal Policy in the Aftermath of 9/11 Jonas Fisher and Martin Eichenbaum	WP-04-06
Merger Momentum and Investor Sentiment: The Stock Market Reaction To Merger Announcements <i>Richard J. Rosen</i>	WP-04-07
Earnings Inequality and the Business Cycle Gadi Barlevy and Daniel Tsiddon	WP-04-08
Platform Competition in Two-Sided Markets: The Case of Payment Networks Sujit Chakravorti and Roberto Roson	WP-04-09
Nominal Debt as a Burden on Monetary Policy Javier Díaz-Giménez, Giorgia Giovannetti, Ramon Marimon, and Pedro Teles	WP-04-10
On the Timing of Innovation in Stochastic Schumpeterian Growth Models <i>Gadi Barlevy</i>	WP-04-11
Policy Externalities: How US Antidumping Affects Japanese Exports to the EU Chad P. Bown and Meredith A. Crowley	WP-04-12
Sibling Similarities, Differences and Economic Inequality Bhashkar Mazumder	WP-04-13

Determinants of Business Cycle Comovement: A Robust Analysis Marianne Baxter and Michael A. Kouparitsas	WP-04-14
The Occupational Assimilation of Hispanics in the U.S.: Evidence from Panel Data <i>Maude Toussaint-Comeau</i>	WP-04-15
Reading, Writing, and Raisinets ¹ : Are School Finances Contributing to Children's Obesity? <i>Patricia M. Anderson and Kristin F. Butcher</i>	WP-04-16
Learning by Observing: Information Spillovers in the Execution and Valuation of Commercial Bank M&As Gayle DeLong and Robert DeYoung	WP-04-17
Prospects for Immigrant-Native Wealth Assimilation: Evidence from Financial Market Participation Una Okonkwo Osili and Anna Paulson	WP-04-18
Individuals and Institutions: Evidence from International Migrants in the U.S. <i>Una Okonkwo Osili and Anna Paulson</i>	WP-04-19
Are Technology Improvements Contractionary? Susanto Basu, John Fernald and Miles Kimball	WP-04-20
The Minimum Wage, Restaurant Prices and Labor Market Structure Daniel Aaronson, Eric French and James MacDonald	WP-04-21
Betcha can't acquire just one: merger programs and compensation <i>Richard J. Rosen</i>	WP-04-22
Not Working: Demographic Changes, Policy Changes, and the Distribution of Weeks (Not) Worked <i>Lisa Barrow and Kristin F. Butcher</i>	WP-04-23
The Role of Collateralized Household Debt in Macroeconomic Stabilization <i>Jeffrey R. Campbell and Zvi Hercowitz</i>	WP-04-24
Advertising and Pricing at Multiple-Output Firms: Evidence from U.S. Thrift Institutions <i>Robert DeYoung and Evren Örs</i>	WP-04-25
Monetary Policy with State Contingent Interest Rates Bernardino Adão, Isabel Correia and Pedro Teles	WP-04-26
Comparing location decisions of domestic and foreign auto supplier plants Thomas Klier, Paul Ma and Daniel P. McMillen	WP-04-27
China's export growth and US trade policy Chad P. Bown and Meredith A. Crowley	WP-04-28
Where do manufacturing firms locate their Headquarters? J. Vernon Henderson and Yukako Ono	WP-04-29
Monetary Policy with Single Instrument Feedback Rules Bernardino Adão, Isabel Correia and Pedro Teles	WP-04-30

Firm-Specific Capital, Nominal Rigidities and the Business Cycle David Altig, Lawrence J. Christiano, Martin Eichenbaum and Jesper Linde	WP-05-01
Do Returns to Schooling Differ by Race and Ethnicity? Lisa Barrow and Cecilia Elena Rouse	WP-05-02
Derivatives and Systemic Risk: Netting, Collateral, and Closeout Robert R. Bliss and George G. Kaufman	WP-05-03
Risk Overhang and Loan Portfolio Decisions Robert DeYoung, Anne Gron and Andrew Winton	WP-05-04
Characterizations in a random record model with a non-identically distributed initial record <i>Gadi Barlevy and H. N. Nagaraja</i>	WP-05-05
Price discovery in a market under stress: the U.S. Treasury market in fall 1998 <i>Craig H. Furfine and Eli M. Remolona</i>	WP-05-06
Politics and Efficiency of Separating Capital and Ordinary Government Budgets Marco Bassetto with Thomas J. Sargent	WP-05-07
Rigid Prices: Evidence from U.S. Scanner Data Jeffrey R. Campbell and Benjamin Eden	WP-05-08
Entrepreneurship, Frictions, and Wealth Marco Cagetti and Mariacristina De Nardi	WP-05-09
Wealth inequality: data and models Marco Cagetti and Mariacristina De Nardi	WP-05-10
What Determines Bilateral Trade Flows? Marianne Baxter and Michael A. Kouparitsas	WP-05-11
Intergenerational Economic Mobility in the U.S., 1940 to 2000 Daniel Aaronson and Bhashkar Mazumder	WP-05-12
Differential Mortality, Uncertain Medical Expenses, and the Saving of Elderly Singles Mariacristina De Nardi, Eric French, and John Bailey Jones	WP-05-13
Fixed Term Employment Contracts in an Equilibrium Search Model Fernando Alvarez and Marcelo Veracierto	WP-05-14
Causality, Causality, Causality: The View of Education Inputs and Outputs from Economics Lisa Barrow and Cecilia Elena Rouse	WP-05-15

Competition in Large Markets Jeffrey R. Campbell	
Why Do Firms Go Public? Evidence from the Banking Industry Richard J. Rosen, Scott B. Smart and Chad J. Zutter	WP-05-17
Clustering of Auto Supplier Plants in the U.S.: GMM Spatial Logit for Large Samples <i>Thomas Klier and Daniel P. McMillen</i>	WP-05-18
Why are Immigrants' Incarceration Rates So Low? Evidence on Selective Immigration, Deterrence, and Deportation <i>Kristin F. Butcher and Anne Morrison Piehl</i>	WP-05-19
Constructing the Chicago Fed Income Based Economic Index – Consumer Price Index: Inflation Experiences by Demographic Group: 1983-2005 Leslie McGranahan and Anna Paulson	WP-05-20
Universal Access, Cost Recovery, and Payment Services Sujit Chakravorti, Jeffery W. Gunther, and Robert R. Moore	WP-05-21
Supplier Switching and Outsourcing Yukako Ono and Victor Stango	WP-05-22
Do Enclaves Matter in Immigrants' Self-Employment Decision? Maude Toussaint-Comeau	WP-05-23
The Changing Pattern of Wage Growth for Low Skilled Workers Eric French, Bhashkar Mazumder and Christopher Taber	WP-05-24
U.S. Corporate and Bank Insolvency Regimes: An Economic Comparison and Evaluation <i>Robert R. Bliss and George G. Kaufman</i>	WP-06-01
Redistribution, Taxes, and the Median Voter Marco Bassetto and Jess Benhabib	WP-06-02
Identification of Search Models with Initial Condition Problems Gadi Barlevy and H. N. Nagaraja	WP-06-03
Tax Riots Marco Bassetto and Christopher Phelan	WP-06-04
The Tradeoff between Mortgage Prepayments and Tax-Deferred Retirement Savings Gene Amromin, Jennifer Huang, and Clemens Sialm	WP-06-05
Why are safeguards needed in a trade agreement? Meredith A. Crowley	WP-06-06

Taxation, Entrepreneurship, and Wealth Marco Cagetti and Mariacristina De Nardi	WP-06-07
A New Social Compact: How University Engagement Can Fuel Innovation Laura Melle, Larry Isaak, and Richard Mattoon	WP-06-08
Mergers and Risk Craig H. Furfine and Richard J. Rosen	WP-06-09
Two Flaws in Business Cycle Accounting Lawrence J. Christiano and Joshua M. Davis	WP-06-10
Do Consumers Choose the Right Credit Contracts? Sumit Agarwal, Souphala Chomsisengphet, Chunlin Liu, and Nicholas S. Souleles	WP-06-11
Chronicles of a Deflation Unforetold François R. Velde	WP-06-12
Female Offenders Use of Social Welfare Programs Before and After Jail and Prison: Does Prison Cause Welfare Dependency? Kristen F. Butcher and Robert J. LaLonde	WP-06-13
Eat or Be Eaten: A Theory of Mergers and Firm Size Gary Gorton, Matthias Kahl, and Richard Rosen	WP-06-14